

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Bucher, Monika; Hauck, Achim; Neyer, Ulrike

Working Paper

Frictions in the interbank market and uncertain liquidity needs: Implications for monetary policy implementation

DICE Discussion Paper, No. 134 [rev.]

Provided in Cooperation with:

Düsseldorf Institute for Competition Economics (DICE), Heinrich Heine University Düsseldorf

Suggested Citation: Bucher, Monika; Hauck, Achim; Neyer, Ulrike (2014): Frictions in the interbank market and uncertain liquidity needs: Implications for monetary policy implementation, DICE Discussion Paper, No. 134 [rev.], ISBN 978-3-86304-133-5, Heinrich Heine University Düsseldorf, Düsseldorf Institute for Competition Economics (DICE), Düsseldorf

This Version is available at: https://hdl.handle.net/10419/100031

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



DISCUSSION PAPER

No 134

Frictions in the Interbank
Market and Uncertain
Liquidity Needs:
Implications for Monetary
Policy Implementation

Monika Bucher, Achim Hauck, Ulrike Neyer

July 2014



IMPRINT

DISCUSSION PAPER

Published by

düsseldorf university press (dup) on behalf of Heinrich-Heine-Universität Düsseldorf, Faculty of Economics, Düsseldorf Institute for Competition Economics (DICE), Universitätsstraße 1, 40225 Düsseldorf, Germany www.dice.hhu.de

Editor:

Prof. Dr. Hans-Theo Normann

Düsseldorf Institute for Competition Economics (DICE)

Phone: +49(0) 211-81-15125, e-mail: normann@dice.hhu.de

DISCUSSION PAPER

All rights reserved. Düsseldorf, Germany, 2014

ISSN 2190-9938 (online) - ISBN 978-3-86304-133-5

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editor.

Frictions in the Interbank Market and Uncertain Liquidity

Needs: Implications for Monetary Policy Implementation

Monika Bucher*

Achim Hauck[†]

Ulrike Never[‡]

July 2014

Abstract

This paper shows that depending on the distribution of banks' uncertain liquidity needs and on how monetary policy is implemented, frictions in the interbank market may reinforce the effectiveness of monetary policy. These frictions imply that with its lending and deposit facilities the central bank has an additional effective instrument at hand to impose an impact on bank loan supply. While lowering the rate on the lending facility has, taken for itself, an expansionary effect, lowering the rate on the deposit facility has a contractionary effect. This result has interesting implications for monetary policy implementation at the zero lower bound.

JEL classification: E52, E58, G21

Keywords: interbank market, monetary policy, monetary policy implementation, zero

lower bound, loan supply

*Heinrich Heine University Düsseldorf Department of Economics, Universitaetsstrasse 1, 40225 Duesseldorf, and Post-Graduate Programme Global Financial Markets, Halle and Jena, Germany, email: monika.bucher@hhu.de.

†Portsmouth Business School, University of Portsmouth, Richmond Building Portland Street, Portsmouth PO1 3DE, UK, email: achim.hauck@port.ac.uk.

[‡]Corresponding author. Heinrich Heine University Düsseldorf, Department of Economics, Universitaetsstrasse 1, 40225 Duesseldorf, Germany, phone: +49/(0)211/81 11511, email: ulrike.neyer@hhu.de.

1 Introduction

The interbank market for overnight loans is important for monetary policy implementation. By steering the interest rate in this market, the central bank aims to influence short-term nominal interest rates, and thereby, through various channels, the price level and maybe aggregate output. In the euro area, the interest rate channel and the credit channel play an important role.¹ Both transmission mechanisms rest on the central bank's ability to influence bank lending.

During the recent financial crisis, euro area interbank markets seized up. This led to concerns about the Eurosystem's ability, or the lack thereof, to actually control bank lending in times of malfunctioning interbank markets, and it triggered a heated debate of whether the transmission mechanism of monetary policy might be impaired. Our paper aims to contribute to this debate by studying in how far frictions in the interbank market for overnight loans influence the impact of monetary policy on bank loan supply and by discussing the implications for monetary policy implementation.

We develop a theoretical model that has two central features. First, it accounts for interbank market frictions. Second, it captures main elements of the Eurosystem's² operational framework. Frictions in the interbank market emerge in the form of transaction costs. We broadly interpret these transaction costs as search costs. Banks must find suitable transaction partners with first, matching liquidity needs and second, a willingness to conclude mutual agreements for trade. The former may be costly as, for example, banks have to split large transactions into small ones to work around credit lines (Bartolini, Bertola, and Prati, 2001). The latter may be costly because lenders in the overnight interbank market are typically unwilling to expose themselves to any counterparty credit risk (Hauck and Neyer, 2013). Consequently, they engage in costly checks of the credit-worthiness of potential borrowers who in turn must provide costly signals of their credit-worthiness.³ The main elements of the Eurosystem's operational framework captured by

¹For respective empirical analyses for the interest rate channel see, for example, Čihák, Harjes, and Stavrev (2009); Angeloni, Kashyap, Mojon, and Terlizzese (2003). In an empirical analysis referring, for example, to the Spanish credit market, Jiménez, Ongena, Peydró, and Saurina (2012) confirm a high relevance of the credit channel.

²The term "Eurosystem" stands for the institution which is responsible for monetary policy in the euro area, namely the ECB and the national central banks in the euro area. For the sake of simplicity, the terms "ECB" and "Eurosystem" are used interchangeably throughout this paper.

³One of the first papers dealing explicitly with interbank market transaction costs is the one by Bartolini, Bertola, and Prati (2001). They argue that interbank market transaction costs are responsible for the relatively high federal funds rate usually observed at the end of a reserve maintenance period. Transaction costs also play a crucial role in Hauck and Neyer (2013). They argue that transaction costs, or participation

our model are the main refinancing operations and the two standing facilities. The main refinancing operations are credit operations with a maturity of one week by which the Eurosystem provides reserves to the banking sector. The two standing facilities, a deposit facility and a lending facility, allow banks to balance their overnight liquidity needs. The interest rates on the facilities form a corridor around the rate on the main refinancing operations with the rate on the deposit facility to be lower and the rate on the lending facility to be higher than the main policy rate.⁴ It is worth mentioning that although our model focuses on the Eurosystem's operational framework, our results apply to other operational frameworks as well, as long as they allow commercial banks to balance uncertain liquidity needs by using a deposit facility and a lending facility offered by the central bank.

The results of our model replicate several stylized facts and imply interesting implications for monetary policy implementation. The replicated stylized facts observed before and during the recent financial crisis are:

- If there are no interbank market frictions, the interbank market rate will equal the main policy rate, reserves provided by the central bank to the banking sector through its main refinancing operations will correspond to the benchmark allotment,⁵ and the standing facilities will not be used. This is exactly what was observed in the euro area before the outbreak of the financial crisis in 2007.
- Introducing interbank market frictions in our model leads to results which are broadly consistent with the stylized facts observed during the financial crisis. The interbank market rate falls below the main policy rate, reserves provided by the central bank exceed the benchmark allotment and the standing facilities are used whereas the use of the deposit facility outweighs by far the use of the lending facility.

From our model results the following implications for monetary policy implementation can be drawn:

costs, can explain several stylized facts observed in the euro area interbank market during the financial crisis. Models explicitly considering a costly search process in the interbank market can be found for example in Furfine (2004) and Ashcraft and Duffie (2007). Furfine analyzes the effectiveness of standing facilities offered by a central bank at reducing the volatility of the overnight interbank rate. Ashcraft and Duffie show how the search process in a decentralized interbank market influences intraday allocation and the pricing of federal funds.

⁴For a detailed description of the Eurosystem's operational framework see European Central Bank (2012).

⁵The Eurosystem's benchmark allotment is generally understood as the allotment in a main refinancing operation that will allow the banks to smoothly fulfil their reserve requirements taking into account future liquidity needs from reserve requirements and autonomous factors. For details see European Central Bank (2014).

- Irrespectively of interbank market frictions, the central bank can influence banks' expected funding costs, and therefore bank loan supply, by changing its main policy rate even if the frictions imply a total interbank market freeze.
- If interbank market frictions are sufficiently high, they will reinforce the effect of a sole change of the main policy rate on bank loan supply. The reinforcing effect increases in the extent of uncertainty about banks' actual liquidity needs.
- The reinforcing effect will be avoided if the central bank changes all its interest rates (rate on its main refinancing operations and on its standing facilities) to the same extent. Obviously, this may not be possible if the rate on the deposit facility is fixed at the zero lower bound.
- If interbank market frictions are sufficiently high, the central bank can use the rates on the facilities as an additional effective monetary policy instrument by changing the width or the asymmetry of the interest rate corridor. A decrease of the rate on the deposit facility corresponds to a contractionary monetary policy. Consequently, using this instrument for conducting a contractionary monetary policy will not be possible if the zero lower bound becomes binding.

To illustrate the main idea behind the implications of our model results, let us assume that the central bank conducts an expansionary monetary policy by lowering solely the rate on its main refinancing operations. Then, borrowing reserves from the main refinancing operations becomes cheaper which implies that also the price for reserves in the interbank market, the interbank rate, decreases. This means that banks' expected marginal funding costs decline which has a positive impact on their loan supply. However, frictions in the form of transaction costs in the interbank market imply that the interbank rate deviates from the central bank's main policy rate. If these costs are that high, that the interbank rate will be already at its lower bound, which is determined by the rate on the central bank's deposit facility, the above described price mechanism will not work anymore. Therefore, borrowing reserves from the central bank's refinancing operations remains to be relatively cheaper as compared with an interbank market loan. As a consequence, banks increase their borrowing from the refinancing operations. The price effect (lower interbank rate) is replaced by a quantity effect (increased borrowing from the refinancing operations). This quantity effect implies that the expansionary effect of the initial monetary policy impulse is reinforced as the sensitivity of banks' funding costs to the monetary

policy impulse is higher. The central bank can steer the extent of the reinforcing effect by changing the rate on its deposit facility.

This rest of this paper is organized as follows. Section 2 presents related literature. Section 3 describes the framework of the model. Sections 4 and 5 derive the optimal behavior of commercial banks. Section 6 discusses the equilibrium of the model. Taking a closer look at this equilibrium in Section 7, we analyze the impact of monetary policy on bank loan supply and discuss the consequences for monetary policy implementation. Section 8 briefly summarizes the paper.

2 Related Literature

Our paper contributes to three strands of literature. The first strand focuses on the influence of monetary policy on bank lending. A huge part of this literature considers asymmetric information in credit markets and argues that these frictions amplify the effects of monetary policy on bank lending and, therefore, on aggregate demand.⁶ This credit view of monetary policy can be divided into the balance sheet channel and the bank lending channel. With respect to the balance sheet channel, the crucial point is that a monetary policy impulse changes the borrowers' net worth. A contractionary monetary policy decreases the borrowers' net worth which implies an increase in adverse selection and moral hazard problems leading to a decline in bank lending. Seminal papers are those by Bernanke and Gertler (1989) and Bernanke, Gertler, and Gilchrist (1999). With respect to the bank lending channel, the driving force is that monetary policy has an impact on bank deposits. A contractionary monetary policy reduces bank deposits implying a decline in bank loan supply. Important papers dealing with this traditional bank lending channel are, for example, those by Gertler and Gilchrist (1993) and Kashyap and Stein (1995). This traditional approach of the bank lending channel has been criticized as it neglects, for example, that banks can replace deposits by market-based funding. However, Disyatat (2011) shows that a greater reliance on market-based funding creates a new approach of the bank lending channel. Crucial is that a stronger reliance on market-based funding increases the sensitivity of banks' funding costs to monetary policy as the banks' health, in terms of leverage, asset quality and in perception of risk, becomes more important.

⁶For a short survey on this so called credit view of monetary policy see, for example, Boivin, Kiley, and Mishkin (2010) and Peek and Rosengreen (2012).

The second strand of literature deals with frictions in the interbank market. Until the outbreak of the financial crisis in 2007, the interbank market was typically regarded as frictionless, also in the theoretical literature. As a consequence, the interbank rate was assumed to be identical with the monetary policy rate or the interbank market was entirely neglected. The financial crisis challenged this view and inspired a growing literature dealing with interbank market imperfections, primarily focusing on asymmetric information about credit risks. Freixas and Jorge (2008) consider the impact of this interbank market friction for the transmission of monetary policy. They show that private information in the interbank market with respect to credit risks may induce rationing of firms in credit markets. With respect to the transmission mechanism of monetary policy this implies that asymmetric information in the interbank market may be responsible for a) a magnitude effect, i.e. the aggregate impact of monetary policy may be large given the small interest elasticity of investment, and b) a liquidity effect, i.e. that the impact of monetary policy is stronger for banks with less liquid balance sheets. Heider, Hoerova, and Holthausen (2009) argue that banks' informational disadvantage with respect to counterparty credit risks induces them to hold more liquidity. Depending on the risk dispersion, this may result in either adverse selection or a dry-up of the interbank market. However, banks may learn about counterparty credit risks by repeatedly trading with each other so that the asymmetric information problem may be mitigated. In an empirical analysis of the German unsecured overnight money market, Bräuning and Fecht (2012) determine the impact of such relationship lending for banks' ability to access liquidity. The causes of a possible dry-up of the interbank market are also analyzed by Allen, Carletti, and Gale (2009). They show that banks will start to hoard liquidity if they are unable to hedge idiosyncratic liquidity shocks.

The third strand of literature deals with monetary policy implementation, bank behavior, and consequences for the conditions in the overnight interbank market. This literature can be divided into three groups. The first group focuses on the U.S. before the outbreak of the financial crisis in 2007. Considering major institutional characteristics of the federal funds market, Ho and Saunders (1985) as well as Clouse and Dow (2002) analyze banks' reserve management and draw conclusions for the conditions in the interbank market for reserves. However, the largest part of the literature dealing with the federal funds mar-

ket focuses on why the federal funds rate fails to follow a martingale within the reserve maintenance period.⁷

The second group of the literature refers to the euro area in the pre-crisis period. A bulk of this literature deals with the under- and overbidding behavior in the Eurosystem's main refinancing operations which could be observed in the first years of the European Monetary Union.⁸ Apart from this, there are papers analyzing the consequences of alternative monetary policy implementations. Nautz (1998) shows that the central bank can influence the interbank market rate by being more or less vague about its future monetary policy. Välimäki (2001) analyzes the effects of alternative tender procedures with respect to the Eurosystem's refinancing operations. Never and Wiemers (2004) refer to the collateral framework. They show that differences in banks' opportunity costs of holding collateral form a rationale for the existence of an interbank market for reserves. Never (2009) demonstrates that remunerating required reserves in a specific way increases the flexibility of monetary policy. Pérez-Quirós and Rodríguez-Mendizábal (2006) show that the two standing facilities offered by the Eurosystem in combination with its minimum reserve system are an effective instrument to stabilize the interbank market rate. Whitesell (2006), although not explicitly referring to the euro area, looks at a minimum reserve system and standing facilities as two alternative regimes for controlling overnight interest rates. Also focusing on the standing facilities, Berentsen and Monnet (2008) develop a general equilibrium framework and show that changing the rates on these facilities may be used actively as a monetary policy instrument. Also Goodhart (2013) points out that by changing the rates on the standing facilities the central bank has an additional instrument at hand. Beaupain and Durré (2008) examine the interday and intraday dynamics of the euro area overnight interbank market and argue that specific features of the Eurosystem's operational framework, as its minimum reserve system, can explain observed regular patterns.

The third group of this third strand of literature comprises papers regarding changes in monetary policy implementation in response to the financial crisis. Eisenschmidt, Hirsch, and Linzert (2009) analyze the relatively aggressive bidding behavior of banks in the Eurosystem's main refinancing operations at the beginning of the financial turmoil. Also

⁷See Hamilton (1996), Clouse and Dow (1999), Furfine (2000), and Bartolini, Bertola, and Prati (2001, 2002).

⁸Under- and overbidding behavior refers to a bidding behavior in which total bids significantly exceed or remain under the Eurosystem's benchmark allotment. Analyses with respect to this under- and overbidding behavior can be found in Ayuso and Repullo (2001, 2003), Ewerhart (2002), Nautz and Oechssler (2003, 2006), and Bindseil (2005).

referring to the first part of the financial crisis (until 2008), Cassola and Huetl (2010) assess the effectiveness of monetary policy implementation during that time. Borio and Disyatat (2009) describe main characteristics of unconventional monetary policies adopted during the financial crisis. They point out that an important feature of these policies is that the central bank also uses its balance sheet to influence prices and conditions in the interbank market. Cheun, von Köppen-Mertes, and Weller (2009) analyze changes to the collateral frameworks of the Eurosystem, the Federal Reserve System and the Bank of England. Lenza, Pill, and Reichlin (2010) describe the way in which these three central banks generally conducted monetary policy during the financial crisis and point to the importance of their influence on money market spreads. Hauck and Neyer (2013) develop a theoretical model considering main institutional features of the Eurosystem's operational framework which has been in place since September 2008 to explain several stylized facts observed during the financial crisis.

Our paper combines all three strands of this literature by analyzing the consequences of frictions in the overnight interbank market, in the form of broadly defined transaction costs, for the impact of monetary policy on bank loan supply. With respect to monetary policy implementation, we point out the crucial role the central bank's standing facilities play for the effectiveness of monetary policy in the presence of interbank market frictions and uncertain liquidity needs.

3 Framework

In our model, we distinguish between three types of agents. We consider a continuum of measure one of price-taking commercial banks with a large number of bank customers and a central bank.

Each commercial bank i grants a loan volume L_i to its customers at a given interest rate i^L . This generates net revenues of

$$i^L L_i - \frac{1}{2} \lambda L_i^2 \tag{1}$$

for bank i. The second term of (1) reflects the costs of managing loans. The quadratic form of this cost function captures the idea that loans differ in their complexity so that the bank adds the least complex loans to its portfolio first.

Bank i credits the loan volume L_i to its customers' demand deposit accounts. The bank's customers can use this newly created money to make payments. They pay by

cash or by transferring deposits. Due to the cash payments, each bank i experiences cash withdrawals cL_i , with the currency ratio c reflecting the share of the newly created money used for cash payments. Due to the payments made by deposit transfers, a share χt_i of the remaining deposits $(1-c)L_i$ is transferred to customers of other banks. Crucial is that the net deposit transfer differs across banks. This is reflected by the bank individual variable t_i . For banks facing a net deposit inflow $t_i < 0$, and for those facing a net deposit outflow $t_i > 0$. For a single bank i, its net deposit transfer is uncertain as t_i is the realization of a random variable T. Across all banks, t_i is distributed in the interval $[t^{min}, t^{max}]$ according to the density function $g(t_i) = G'(t_i)$. Note that

$$E[T] = \int_{t^{min}}^{t^{max}} t_i g(t_i) dt_i = 0.$$
 (2)

The parameter χ with $\chi \in (0, \frac{1}{t^{max}}]$ is a scale parameter which determines the dispersion of the distribution of T.⁹ If χ increases, the distribution will exhibit a higher dispersion. Accordingly, we will use χ as a measure for uncertainty about a bank's net deposit transfer. Considering both, cash withdrawals and the net deposit transfer, bank i's remaining deposits are given by

$$D_i = L_i - cL_i - (1 - c)\chi t_i L_i. (3)$$

This implies that a single bank i may face a liquidity surplus or deficit. Bank i can balance its individual liquidity needs by transacting with the central bank or in the interbank market. However, the cash withdrawals imply that the banking sector as a whole faces a structural liquidity deficit which can only be covered by the central bank being the monopoly producer of currency.

To obtain liquidity from the central bank, each bank i can participate in the central bank's refinancing operations and borrow the amount RO_i at the rate i^{RO} . Moreover, it can use a lending facility to borrow LF_i at the rate i^{LF} . However, it can also place an

⁹As χt_i as a *share* cannot exceed one, χ is restricted to $\frac{1}{t^{max}}$.

 $^{^{10}}$ Generally, credit operations with the central bank require adequate collateral. In our setting a bank's loan volume L_i serves as collateral, and therefore, limits its central bank borrowing. The central bank may impose a haircut on these loans when accepting them as collateral, like in Bindseil and König (2011). In this setting, however, we assume that such a hair cut is not binding and neglect the collateralization of central bank loans. See in this context also our remarks on this aspect made in the introduction.

amount DF_i of liquidity in a deposit facility offered by the central bank at the rate i^{DF} . Accordingly, transactions with the central bank imply net costs of

$$i^{RO}RO_i + i^{LF}LF_i - i^{DF}DF_i. (4)$$

The rates on the facilities form a corridor around the rate on the refinancing operations with $i^{LF} > i^{RO} > i^{DF}$.

A single bank can also borrow and lend liquidity in the interbank market. Bank i's position in this market is B_i . If $B_i > 0$, bank i will borrow the amount B_i at the rate i^{IBM} . Conversely, $B_i < 0$ indicates that bank i will lend the amount $|B_i|$ at this rate. Independently of whether bank i borrows or lends in the interbank market, transaction costs $\gamma |B_i|$ accrue, with $\gamma \geq 0$. Therefore, net costs in the interbank market account for

$$i^{IBM}B_i + \gamma |B_i|. (5)$$

Considering the described costs and revenues, each bank i aims to maximize its profit Π_i . Banks are risk neutral so that by combining equations (1), (4), and (5), the objective function of bank i simply reads

$$\Pi_{i} = i^{L}L_{i} - \frac{1}{2}\lambda L_{i}^{2} - i^{RO}RO_{i} - i^{LF}LF_{i} + i^{DF}DF_{i} - i^{IBM}B_{i} - \gamma |B_{i}|$$
(6)

s.t.
$$L_i + DF_i = RO_i + LF_i + D_i + B_i, \tag{7}$$

where (7) describes bank i's balance sheet constraint. The assets consist of bank i's loans L_i and its deposits held at the central bank DF_i . Its liabilities comprise its central bank borrowing, RO_i+LF_i , and its customers' deposits D_i . The bank's position in the interbank market B_i might constitute an item on the asset or liabilities side of the balance sheet, depending on whether the bank borrows from or lends in the interbank market.

When solving this optimization problem, we have to consider the sequence of moves. First, bank i decides on its optimal loan supply L_i and its optimal borrowing from the central bank's refinancing operations RO_i . When making these decisions, the bank cannot observe the realization t_i of the random variable T. The bank thus faces uncertainty about its future liquidity needs. After bank i observed t_i so that uncertainty about its liquidity needs is resolved, it decides on its transactions in the interbank market B_i and on its use of the central bank's facilities DF_i and LF_i . This sequence of moves implies that the optimization problem can be split up into two stages. Solving this optimization problem

by backward induction, we first investigate the second stage of the model and determine a bank's optimal behavior in the interbank market and its optimal use of the central bank's facilities. Then, we analyze the first stage of the model and determine a bank's optimal lending to the non-banking sector and its optimal borrowing from the central bank's refinancing operations.

4 Optimal Behavior at the Second Stage

At the second stage each bank learns its respective share of transferred deposits t_i and, therefore, its actual liquidity needs. Accordingly, banks face no uncertainty at this stage. Using (3) we define bank i's actual individual liquidity needs as

$$N_i := L_i - RO_i - D_i = L_i (c + (1 - c)\chi t_i) - RO_i.$$
(8)

If $N_i \geq 0$, bank i will inherit a liquidity deficit. In this case, the bank compares marginal costs of borrowing from the interbank market given by $i^{IBM} + \gamma$ with those of using the lending facility which are simply i^{LF} . As the two marginal costs are constant, the bank will cover its total liquidity deficit by borrowing from the lending facility if $i^{IBM} + \gamma > i^{LF}$. If $i^{IBM} + \gamma < i^{LF}$, it will borrow from the interbank market only. In case both marginal costs are identical, the bank is essentially indifferent between interbank borrowing and the usage of the lending facility.

If $N_i < 0$, bank *i* will inherit a liquidity surplus so that the bank decides analogously. If the marginal revenues in the interbank market $i^{IBM} - \gamma$ are higher (lower) than the marginal revenues of the central bank's deposit facility i^{DF} , it will place its total surplus in the interbank market (in the deposit facility). In case marginal revenues are identical, the bank will again be indifferent.

As banks will only trade liquidity in the interbank market if this is more beneficial than using the central bank's facilities, the interbank rate in equilibrium will be $i^{IBM*} \in [i^{DF} + \gamma, i^{LF} - \gamma]$. Whether banks prefer the interbank market thus crucially depends on the magnitude of γ . If transaction costs are too high, each bank will prefer to use the central bank's facilities instead of trading in the interbank market. As a result, the interbank market breaks down. This will be the case if $i^{IBM*} - \gamma < i^{DF}$ and $i^{IBM*} + \gamma > i^{LF}$, i.e. if

$$\gamma > \frac{i^{LF} - i^{DF}}{2} =: \bar{\bar{\gamma}}. \tag{9}$$

We thus obtain

Proposition 1: If $\gamma \leq \bar{\gamma}$, the interbank market will be active and we will have to distinguish between three cases regarding the interbank rate:

$$i^{IBM*} = i^{LF} - \gamma$$
 if $RO < cL$,
 $i^{IBM*} \in [i^{DF} + \gamma, i^{LF} - \gamma]$ if $RO = cL$,
 $i^{IBM*} = i^{DF} + \gamma$ if $RO > cL$. (10)

If $\gamma > \bar{\gamma}$, the interbank market will be inactive.

Proof: Omitted.

The proposition states that the interbank rate depends crucially on the aggregate liquidity position of the banking sector. Denoting aggregate borrowing from the refinancing operations by RO and aggregate lending to the non-banking sector by L, an aggregate liquidity deficit will arise if banks' cash withdrawals cL are larger than the aggregate amount obtained in the refinancing operations RO. In this case, competition for scarce liquidity brings the interbank rate to its upper limit $i^{LF} - \gamma$. A higher interest rate would not be accepted by the liquidity deficit banks, since then they would prefer to borrow from the central bank's lending facility instead. If an aggregate liquidity surplus occurs, as cash withdrawals are lower than the aggregate amount of liquidity obtained in the refinancing operations, competition for limited lending possibilities in the interbank market brings the interbank rate to its lower limit $i^{DF} + \gamma$. If there is neither an aggregate liquidity deficit nor surplus, neither market side possesses market power. In consequence, any rate within the lower and the upper limit depicts a possible equilibrium.

5 Optimal Behavior at the First Stage

5.1 A Bank's Optimization Problem

At the first stage of the model, bank i must decide on its loan volume L_i and on its borrowing from the refinancing operations offered by the central bank RO_i . The decision problem of bank i that aims to maximize its expected profit $E[\pi_i]$ reads

$$\max_{L_{i},RO_{i}} E\left[\pi_{i}\right] = i^{L} L_{i} - \frac{1}{2} \lambda L_{i}^{2} - i^{RO} RO_{i} - \max\left\{i^{IBM} - \gamma, i^{DF}\right\} \int_{t^{min}}^{\bar{t}_{i}} N_{i} g(t_{i}) dt_{i} \\
- \min\left\{i^{IBM} + \gamma, i^{LF}\right\} \int_{\bar{t}_{i}}^{t^{max}} N_{i} g(t_{i}) dt_{i}. \tag{11}$$

The first term on the right hand side of (11) reflects the interest revenues of granting loans to the non-banking sector. The second term expresses the management costs associated with these loans. The last three terms reflect expected funding costs. Funding costs occur as due to certain cash withdrawals and the uncertain net deposit transfer there may be a loss of non-interest bearing deposits. The third term on the right hand side of (11) reflects the certain first-stage funding costs. They accrue due to bank i's borrowing from the central bank's refinancing operations. The last two terms show expected second-stage funding costs. Funding needs at the beginning of the second stage are given by (8). The currency ratio c and the scale parameter χ are certain and identical for all banks. In addition, the amounts L_i and RO_i are certain once they are chosen at the first stage. Therefore, the share of net deposit transfer t_i is the only source of uncertainty of bank i at the first stage regarding its funding needs N_i . From (8) we can infer that bank i will face neither a liquidity deficit nor a surplus at the second stage, i.e. $N_i = 0$, if

$$t_i = \frac{RO_i - cL_i}{(1 - c)\chi L_i} =: \bar{t}_i. \tag{12}$$

It follows directly from (12) that bank i's critical share of net deposit transfer \bar{t}_i increases in RO_i and decreases in L_i . If $(t_i < \bar{t}_i)$, bank i faces a liquidity surplus at the beginning of the second stage $(N_i < 0)$. Considering the distribution of T, the bank's expected liquidity surplus is given by $\int_{t^{min}}^{\bar{t}_i} N_i g(t_i) dt_i / G(\bar{t}_i)$. The bank will lend its excess liquidity in the interbank market or will place it in the deposit facility, depending on which option is more profitable. Consequently, expected revenues in case of a liquidity surplus are

$$-\max\left\{i^{IBM} - \gamma, i^{DF}\right\} \frac{\int_{t^{min}}^{\overline{t}_i} N_i g(t_i) dt_i}{G(\overline{t}_i)}.$$

Analogously, expected costs in case of a liquidity deficit are

$$\min \{i^{IBM} + \gamma, i^{LF}\} \frac{\int_{\bar{t}_i}^{t^{max}} N_i g(t_i) dt_i}{1 - G(\bar{t}_i)}.$$

As the former case occurs with probability $G(\bar{t}_i)$, and the latter with probability $1 - G(\bar{t}_i)$, the last two terms of (11) show expected second-stage funding costs resulting from transactions in the interbank market or from using the central bank's facilities.

Note that uncertainty with respect to the share of net deposit transfer exists only at the individual level. At the aggregate level, the net deposit transfer must be zero so that E[T] = 0 (see equation (2)). This has the following implications. First, for any given loan volume L_i and any amount RO_i of liquidity obtained in the refinancing operations, each bank has the same expectations about its subsequent liquidity needs. Second, banks form the same expectations about the interbank rate that will prevail in equilibrium. The interbank rate will only depend on the aggregate liquidity position in the banking sector. Once all banks have granted their loans and borrowed from the refinancing operations, this aggregate liquidity position is certain. This implies that an individual bank takes the aggregate liquidity needs and, therefore, also the interbank rate as given. Consequently, all banks face exactly the same decision problem given by (11). The optimal individual borrowing from the refinancing operations RO_i^{opt} as well as the optimal individual lending to the non-banking sector L_i^{opt} are identical for all banks and are, therefore, equal to the respective aggregate values RO and L. The same applies for the critical share of net deposit transfer \bar{t}_i^{opt} .

5.2 Optimal Borrowing from the Refinancing Operations

Determining a bank's optimal behavior at the first stage, we can restrict our attention to the case $i^{IBM} \in [i^{RO} - \gamma, i^{RO} + \gamma]$. Suppose $i^{IBM} < i^{RO} - \gamma$. Then, no bank has an incentive to borrow from the central bank's refinancing operations at the first stage since borrowing from the interbank market at the second stage is strictly cheaper. However, refusing to borrow from the central bank's refinancing operations at the first stage implies that there is no liquidity to be traded on an interbank market so that the interbank market is inexistent. Therefore, $i^{IBM} < i^{RO} - \gamma$ constitutes no possible equilibrium. If $i^{IBM} > i^{RO} + \gamma$ each bank would be incentivized to borrow unlimitedly from the central bank's refinancing operations to place its liquidity in the interbank market. Apparently, this cannot be an equilibrium either. Considering this and solving the optimization problem (11) we obtain

Lemma 1: Suppose that $i^{IBM*} \in \left[i^{RO} - \gamma, i^{RO} + \gamma\right]$. If $\overline{t}_i^{opt} \geq \tilde{t} := -\frac{c}{(1-c)\chi}$, then at the first stage, bank i will borrow from the central bank's refinancing operations according to:

$$i^{RO} = \max\left\{i^{IBM*} - \gamma, i^{DF}\right\} G\left(\overline{t}_{i}^{opt}\right) + \min\left\{i^{IBM*} + \gamma, i^{LF}\right\} \left[1 - G\left(\overline{t}_{i}^{opt}\right)\right]. \tag{13}$$

Proof: See appendix.

Optimal borrowing from the central bank's refinancing operations requires marginal costs of this borrowing to be equal to expected marginal revenues. Marginal costs are equal to the interest rate on these operations given by the left hand side of (13). The right

hand side of (13) reflects expected marginal revenues. With probability $G(\bar{t}_i^{opt})$, bank i will face a liquidity surplus at the second stage, i.e. $N_i < 0$. In this case, the bank will either lend its excess liquidity in the interbank market or place it in the deposit facility, depending on which alternative yields the higher revenues. With probability $1 - G(\bar{t}_i^{opt})$, the bank will face a liquidity deficit, i.e. $N_i > 0$, so that it will borrow from the interbank market or close its liquidity gap by borrowing from the central bank's lending facility. In this case borrowing from the central bank's refinancing operations implies marginal revenues in the form of avoided illiquidity costs.

The adjustment process in the case marginal costs differ from expected marginal revenues plays a crucial role in our analysis. This process can be described as follows. Assume that marginal costs are higher than expected marginal revenues. Then, the bank will reduce its borrowing from the refinancing operations. This increases the probability $G(\bar{t}_i)$ of facing a liquidity deficit at the second stage. As marginal revenues in the case of a liquidity deficit given by min $\{i^{IBM} + \gamma, i^{LF}\}$ are strictly larger than those in the case of a liquidity surplus given by max $\{i^{IBM} - \gamma, i^{DF}\}$, expected marginal revenues will increase if the bank reduces its borrowing from the refinancing operations. Consequently, as long as marginal costs are higher than marginal revenues, the bank wishes to reduce RO_i until expected marginal revenues equal marginal costs.

However, if the non-negativity constraint on RO_i becomes binding, the bank cannot reduce RO_i any further. This will be the case if $\bar{t}_i^{opt} < \tilde{t}$. As a result, \bar{t}_i^{opt} can not be realized by bank i and $RO_i^{opt} = 0$. Accordingly, marginal costs of borrowing from the refinancing operations remain higher than expected marginal revenues.

5.3 Optimal Lending to the Non-Banking Sector

Solving the optimization problem (11) with respect to the optimal lending L_i^{opt} to the non-banking sector, we obtain

Lemma 2: If $\bar{t}_i^{opt} \geq \tilde{t} := -\frac{c}{(1-c)\chi}$, bank i will supply loans at the first stage according to the following first order condition:

$$i^{L} = \lambda L_{i}^{opt} + ci^{RO} + (1 - c)\chi \max\left\{i^{IBM} - \gamma, i^{DF}\right\} E\left[t_{i}|t_{i} < \overline{t}_{i}^{opt}\right] G\left(\overline{t}_{i}^{opt}\right) + (1 - c)\chi \min\left\{i^{IBM} + \gamma, i^{LF}\right\} E\left[t_{i}|t_{i} > \overline{t}_{i}^{opt}\right] \left[1 - G\left(\overline{t}_{i}^{opt}\right)\right],$$

$$(14)$$

¹¹A decrease in RO_i reduces \bar{t}_i , as equation (12) reveals.

 $^{^{12}\}text{If }RO_{i}^{opt}=0\text{, then }\overline{t}_{i}^{opt}=-\frac{c}{(1-c)\chi}=:\widetilde{t}.$

with \bar{t}_i^{opt} being implicitly defined by (13).

Proof: See appendix.

Optimal lending L_i^{opt} to the non-banking sector requires balancing marginal revenues with expected marginal costs of granting loans. Marginal revenues are equal to the interest rate i^L . Expected marginal costs consist of marginal management costs λL_i^{opt} and expected marginal funding costs. The latter can be divided into two parts. The first part ci^{RO} refers to the certain funding costs due to borrowing from the central bank's refinancing operations at the first stage. The second part corresponds to the expected second stage marginal funding costs.

A liquidity surplus materializes with probability $G(\bar{t}_i^{opt})$ and a liquidity deficit with probability $1 - G(\bar{t}_i^{opt})$. If bank i faces a liquidity surplus at the beginning of the second stage as $t_i \leq \bar{t}_i^{opt}$, it will either lend its liquidity surplus in the interbank market at $i^{IBM} - \gamma$ or place it in the deposit facility at i^{DF} . Hence, the third term on the right hand side of (14) determines the expected (negative) marginal funding costs in the case of a liquidity surplus. The fourth term determines the expected marginal funding costs in the case of a liquidity deficit. A liquidity deficit will occur if $t_i > \bar{t}_i^{opt}$. In this case, bank i will borrow either from the interbank market at $i^{IBM} + \gamma$ or from the lending facility at i^{LF} .

With respect to our comparative static analysis, it is useful to rewrite expected second stage marginal funding costs as follows

$$(1-c)\chi E\left[t_i|t_i > \bar{t}_i^{opt}\right] \left[1 - G\left(\bar{t}_i^{opt}\right)\right] 2\gamma \qquad \text{if } \gamma \leq \bar{\bar{\gamma}}, \tag{15}$$

$$(1-c)\chi E\left[t_i|t_i>\bar{t}_i^{opt}\right]\left[1-G\left(\bar{t}_i^{opt}\right)\right]\left(i^{LF}-i^{DF}\right) \qquad \text{if } \gamma>\bar{\bar{\gamma}}. \tag{16}$$

Equation (15) and (16) reveal that expected second stage marginal funding costs consist of the expected share per unit of loans for which funding costs are expected to accrue and the relevant funding costs. For details with respect to a bank's expected second stage marginal funding costs we refer the reader to Appendix A.3.

6 Equilibrium

After having clarified the optimal behavior of an individual bank, we are now in a position to determine the equilibrium of our model. We have a continuum of ex-ante identical banks of unit mass. Consequently, the bank-individual optimal values L_i^{opt} and RO_i^{opt} correspond to the respective equilibrium aggregate levels L^* and RO^* .

In the euro area, aggregate borrowing from the ECB's main refinancing operations has been systematically equal to or higher than the ECB's benchmark allotment. 13 As in our model aggregate cash withdrawals cL^* corresponds to the benchmark allotment, we focus in our analysis on equilibria in which $RO^* \geq cL^*$. These equilibria will emerge if G(0) < 0.5and if $i^{LF} - i^{RO} > i^{RO} - i^{DF}$. The latter means that the corridor, which the rates on the central bank's facilities form around the main policy rate, is symmetric or asymmetric in the sense that the lower difference is smaller than the upper difference. These two cases have been relevant in the euro area. 14 The former implies a left-skewed distribution of T so that for each single bank the probability of facing a net deposit outflow is larger than the probability of a net deposit inflow. Due to E[T] = 0, each individual bank thus expects small outflows with a high probability and large inflows with low probability. To understand this pattern, it is useful to distinguish between two types of bank customers. First, each bank possesses a huge number of bank customers predominantly generating relatively small deposit outflows, such as households who make payments for consumption purposes. Second, each bank possesses a small number of bank customers predominantly receiving relatively large payments, such as firms as suppliers of consumption goods. With a small probability these firms may benefit from spikes in demand for their goods, e.g. caused by a major innovation. In consequence, their respective bank would face a large net deposit inflow. Moreover, a large net deposit inflow may also occur if outflows decline to a large extent. This is the case, if households experience a massive shock that significantly reduces their spending, e.g. if in a region cash card payments are not feasible. As the probability of such a shock is also small, banks expect a large net deposit inflow with a small probability and small net deposit outflows with a large probability.

Combining the results of Proposition 1, Lemma 1 and Lemma 2, we obtain

Proposition 2: Assume that G(0) < 0.5 and that $i^{LF} - i^{RO} \ge i^{RO} - i^{DF}$. Then, depending on γ , we have to distinguish between three equilibria

I:
$$RO^* = cL^*$$
, $DF^* = 0$, $LF^* = 0$, $i^{IBM*} = i^{RO} - \gamma [1 - 2G(0)]$ if $\gamma \le \bar{\gamma}$, (17)

II:
$$RO^* > cL^*$$
, $DF^* > 0$, $LF^* = 0$, $i^{IBM*} = i^{DF} + \gamma$ if $\gamma \in (\bar{\gamma}, \bar{\bar{\gamma}}]$, (18)

III:
$$RO^* > cL^*$$
, $DF^* > LF^* > 0$ if $\gamma > \bar{\gamma}$, (19)

 $^{^{13}}$ See footnote 5 for a description of the benchmark allotment. Note that in our model there are no reserve requirements and autonomous factors consists of currency holdings.

¹⁴From April 1999 until November 2013 the rates on the Eurosystem's facilities generally formed a symmetric corridor around the rate on the main refinancing operations. However, due to the zero lower bound, the rate on the deposit facility was not decreased in November 2013 contrary to the rates on the lending facility and on the main refinancing operations. Consequently, there has been an asymmetric interest rate corridor since then.

with

$$\bar{\gamma} := \frac{i^{RO} - i^{DF}}{2[1 - G(0)]},\tag{20}$$

and $\bar{\gamma}$ being defined in (9).

Proof: See appendix.

In Equilibrium I, interbank market transaction costs are that low that each bank borrows exactly an amount equal to its cash withdrawals from the central bank's refinancing operations and balances its liquidity needs resulting from the deposit transfers of its customers solely by using the interbank market. None of the facilities is used. Only this behavior implies that the optimality condition given by (13) is fulfilled. To see this, suppose as a starting point that $\gamma = 0$. If banks borrowed an amount larger than their cash withdrawals from the refinancing operations, there would be an aggregate surplus at the second stage bringing the interbank market to its lower bound $i^{DF} + \gamma = i^{DF}$. However, this cannot be an equilibrium, as then marginal costs of borrowing from the refinancing operations given by i^{RO} would exceed expected marginal revenues which in this case are equal to i^{DF} . Consequently, banks will have an incentive to reduce their borrowing from the refinancing operations to balance marginal costs and expected marginal revenues (see the adjustment process described in Subsection 5.2). Analogously, if banks borrowed an amount lower than their cash withdrawals, there would be an aggregate liquidity deficit bringing the interbank rate to its upper bound $i^{LF} - \gamma = i^{LF}$. This is no equilibrium either, as for this interbank rate expected marginal revenues of borrowing from the refinancing operations, which are equal to i^{LF} , exceed marginal costs given by i^{RO} , so that banks wish to increase their borrowing. Consequently, for $\gamma = 0$ an equilibrium will be reached if each bank borrows an amount equal to its cash withdrawals from the refinancing operations. This implies that the interbank market rate i^{IBM*} equals the central bank's policy rate i^{RO} .

Let us assume next that γ becomes positive. Then, expected marginal revenues of borrowing from the central bank's refinancing operations increase. If a bank faces a liquidity surplus at the second stage, its marginal revenues from placing liquidity in the interbank market will decrease but its avoided marginal illiquidity costs in case of a liquidity deficit will increase (see Lemma 1). Due to the left-skewed distribution of T, the probability of facing a liquidity deficit at the second stage will be higher than of facing a liquidity

¹⁵One obtains expected marginal revenues by inserting the equilibrium interbank market rate into the right hand side of (13).

surplus as long as each bank borrows an amount equal to its cash withdrawals from the central bank's refinancing operations, i.e. as long as $RO_i = cL_i$. Consequently, a bank's expected marginal revenues of borrowing from the refinancing operations will increase if γ becomes positive. To balance marginal costs and expected marginal revenues again, each bank will have an incentive to increase its borrowing from the refinancing operations above cL_i (see the adjustment process described in Subsection 5.2). However, such an aggregate borrowing behavior will result in excess aggregate liquidity so that the interbank rate will decline. This decline will reduce expected marginal revenues of borrowing from the refinancing operations. Accordingly, the incentive to borrow more than cL_i becomes weaker. No bank will borrow more than an amount equal to its cash withdrawals from the refinancing operations if the interbank rate decreases to $i^{IBM} = i^{RO} - \gamma \left[1 - 2G(0)\right]$. As long as the interbank rate is unrestricted, an increase in expected marginal revenues due to higher interbank market transaction costs will thus be offset by a decrease of the interbank rate. The price mechanism works. This mechanism ensures that banks have no incentive to borrow more than an amount equal to their cash withdrawals from the central bank's refinancing operations. Equilibrium I as described by equation (17) will be realized.

If transaction costs exceed the critical level $\bar{\gamma}$, further decreases of the interbank rate will not be possible to balance marginal costs and expected marginal revenues as the interbank rate has reached its lower bound $i^{DF} + \gamma$. Consequently, a further increase in γ implies that banks actually start to increase their borrowing from the refinancing operations. This will reduce the probability of facing a liquidity deficit and, therefore, expected marginal revenues. As the price mechanism does not function, marginal costs and expected marginal revenues are balanced via a quantity effect. As this behavior implies that banks increase their borrowing from the refinancing operations above the cash withdrawals, an aggregate liquidity surplus will materialize, $RO^* > cL^*$, while the interbank rate will remain at its lower limit. The excess liquidity will then be placed in the deposit facility. Hence, for sufficiently high transaction costs, Equilibrium II given in equation (18) will be realized. In this equilibrium, all banks with a liquidity deficit still cover their liquidity needs by using the interbank market. Some surplus banks have to use the deposit facility due to the aggregate excess liquidity.

If transaction costs reach the critical level $\bar{\gamma}$, the deficit banks are no longer willing to borrow their liquidity from the interbank market but prefer to use the central bank's lending facility instead. The interbank market breaks down. Both, the deficit banks as

well as the surplus banks, exclusively use the facilities to balance their liquidity needs at the second stage. As there is aggregate excess liquidity, it follows that $DF^* > LF^*$. Equilibrium III as given in equation (19) will be realized.

In Proposition 2 we assume that G(0) < 0.5 and that $i^{LF} - i^{RO} \ge i^{RO} - i^{DF}$. These assumptions imply that the equilibrium is characterized by $RO^* \ge cL^*$, which is the situation observed in the euro area. However, for the sake of completeness, we will briefly comment on the possible equilibrium $RO^* < cL^*$. This equilibrium will emerge if the distribution of T becomes sufficiently right-skewed or if the interest rate corridor becomes sufficiently asymmetric with $i^{LF} - i^{RO} < i^{RO} - i^{DF}$. We start with the importance of the distribution of T. Let us assume that T is distributed symmetrically around $t_i = 0$ and that the interest rates on the facilities form a symmetric corridor around the main policy rate. Then, the probability of facing a net deposit outflow equals the probability of facing a net deposit inflow due to customers' deposit transfers. In this case, only $RO_i = cL_i$ implies that (13) is fulfilled. Transaction costs play no role as they increase marginal revenues of borrowing from the refinancing operations in the case of a liquidity deficit and decrease them in the case of a liquidity surplus, and for $RO_i = cL_i$ both scenarios occur with the same probability given the symmetric distribution of T.

Now let us assume that the distribution of T becomes right-skewed. For $RO_i = cL_i$, the probability of facing a liquidity surplus increases. In this case, transaction costs imply that expected marginal revenues of borrowing from the refinancing operations decrease. Accordingly, banks are incentivized to borrow less from the refinancing operations. Analogously to the case of a left-skewed distribution of T this results in an increase in the interbank market rate to balance marginal costs and expected marginal revenues of borrowing from the refinancing operations. However, if the interbank rate reaches its upper bound so that it cannot increase further, banks start to borrow less from the refinancing operations and $RO^* < cL^*$.

In order to highlight the importance of the asymmetry of the interest rate corridor let us assume that net deposit transfers T are distributed symmetrically around zero. If then the asymmetry of the interest rate corridor is, contrary to our assumption in Proposition 2, given by $i^{LF} - i^{RO} < \gamma < i^{RO} - i^{DF}$, the optimal behavior of bank i will be $RO_i^{opt} < cL_i^{opt}$: Facing relatively high interbank market transaction costs $(i^{LF} - i^{RO} < \gamma)$, banks with a liquidity deficit will only accept an interbank rate below i^{RO} . Otherwise, they would prefer to use the lending facility. However, an interbank rate below i^{RO} implies for $RO_i = cL_i$ expected marginal revenues of borrowing from the refinancing operations to be lower than

marginal costs.¹⁶ Therefore, banks are incentivized to borrow less from the refinancing operations. Generally, this would lead to an increase in the interbank rate balancing expected marginal revenues and marginal costs again. However, such an adjustment is not possible as in case of a liquidity deficit a bank would not accept a higher interbank rate. Therefore, all banks actually start to borrow less from the refinancing operations to balance expected marginal revenues so that marginal costs and $RO^* < cL^*$.

7 Monetary Policy and Bank Loan Supply

This section analyzes the impact of monetary policy on bank loan supply. As the strength of this effect depends on the extent of uncertainty about the net deposit transfer and interbank market transactions costs, we will start with a brief look on their direct impact on bank loan supply.

7.1 Uncertainty and Interbank Market Frictions

Uncertainty about the net deposit transfer and interbank market frictions have a negative impact on bank loan supply.¹⁷ Both, uncertainty and interbank market transaction costs, influence expected second stage marginal funding costs as Lemma 2 reveals. An increase in uncertainty about the net deposit transfer leads to an increase in both, the expected liquidity surplus and the expected liquidity deficit per unit of loans as revealed by the equations (15) and (16). As this surplus or deficit has to be costly balanced either in the interbank market or via the central bank's facilities, an increase in uncertainty implies higher expected marginal funding costs of granting loans. Consequently, banks start to reduce their loan supply.

Obviously, a change in interbank market transaction costs will only have an impact on bank loan supply if these costs are still low enough to ensure an active interbank market, i.e. if $\gamma \leq \bar{\gamma}$. In this case, it follows from Lemma 2 that an increase in marginal transaction costs γ results in an increase in expected marginal funding costs. Accordingly, banks will reduce their loan supply.

¹⁶For $RO_i^{opt} = cL_i^{opt}$ we get that $\bar{t}_i = 0$ as equation (12) shows. This means for a symmetric distribution of T that $G(\bar{t}_i = 0) = 0.5$, so that expected marginal revenues of borrowing from the refinancing operations are equal to the interbank market rate as revealed by the right hand side of equation (13). As marginal costs of borrowing from the refinancing operations are given by i^{RO} , $i^{IBM} < i^{RO}$ implies expected marginal revenues to be lower than marginal costs.

¹⁷We provide the respective formal analysis in Appendix A.4.

7.2 Main Monetary Policy Rate

The central bank has several alternatives to impose an effect on bank loan supply. Starting with the possibility of changing its policy rate i^{RO} , we obtain

Proposition 3: A change in the policy rate implies for Equilibrium j

$$\begin{split} &\frac{\partial L^{j*}}{\partial i^{RO}} = -\frac{1}{\lambda} \left[c + (1-c)\chi \overline{t}^* \right] < 0 \quad \forall \quad j \quad with \\ &\frac{\partial L^{\text{III}*}}{\partial i^{RO}} < \frac{\partial L^{\text{II}*}}{\partial i^{RO}} < \frac{\partial L^{\text{II}*}}{\partial i^{RO}} < 0. \end{split}$$

Furthermore, we get

$$\begin{split} \frac{\partial^2 L^{\text{III}*}}{\partial i^{RO}\partial\chi} < & \frac{\partial^2 L^{\text{II}*}}{\partial i^{RO}\partial\chi} < & \frac{\partial^2 L^{\text{I}*}}{\partial i^{RO}\partial\chi} = 0, \\ \frac{\partial^2 L^{\text{III}*}}{\partial i^{RO}\partial\gamma} = 0, & \frac{\partial^2 L^{\text{II}*}}{\partial i^{RO}\partial\gamma} < 0, & \frac{\partial^2 L^{\text{I}*}}{\partial i^{RO}\partial\gamma} = 0. \end{split}$$

Proof: See appendix.

This proposition reveals that, independently of potential frictions in the interbank market and the extent of uncertainty about the net deposit transfer, by changing its policy rate i^{RO} the central bank affects banks' marginal funding costs and, therefore, their loan supply in all equilibria. However, frictions in the interbank market and uncertainty about the net deposit transfer may reinforce the impact of this monetary policy impulse on bank loan supply.

In Equilibrium I, in which $\bar{t}^*=0$, the effect of monetary policy on aggregate loan supply depends only on the management cost parameter λ and the currency ratio c. The impact of monetary policy on bank loan supply is not influenced by interbank market frictions or by uncertainty about the net deposit transfer. An increase in i^{RO} implies marginal costs of borrowing from the main refinancing operations to become higher than expected marginal revenues, leading to the adjustment process described in Subsection 5.2. The interbank rate increases which again balances marginal costs and expected marginal revenues. In this case, only the price effect prevails. A quantity effect to balance marginal costs and expected marginal revenues does not occur so that \bar{t}^* remains unchanged. Therefore, we can conclude from Lemma 2 that in Equilibrium I only first stage marginal costs of granting loans will change if the monetary policy rate is changed. Frictions in the interbank market and uncertainty about the net deposit transfer, which both refer to the second stage, play no role for the effectiveness of monetary policy.

In contrast, in Equilibrium II and III a change in the policy rate i^{RO} does not only have an impact on first stage marginal funding costs but also on expected second stage marginal funding costs, i.e. also on the third term given on the right hand side of (14). The reason is that in both equilibria, a change in the interbank market rate to balance marginal costs and expected marginal revenues of borrowing from the central bank's refinancing operations is not feasible. Either the interbank market rate is at its lower bound or an interbank market does not exist due to high transaction costs. 18 Consequently, banks start to adjust their borrowing from the refinancing operations to balance marginal costs and expected marginal revenues. According to Lemma 2, this behavior implies a change in expected second stage marginal funding costs. Suppose the central bank decreases i^{RO} . Then, marginal costs of borrowing from the refinancing operations fall below expected marginal revenues inducing banks to increase their borrowing from the refinancing operations. This borrowing behavior implies an increase in \bar{t}_i^{opt} . The expected liquidity deficit per unit of loans, which has to be covered costly by borrowing from the interbank market (Equilibrium II) or from the central bank's lending facility (Equilibrium III), decreases, and therefore, also banks' expected second marginal funding costs (see equations (15) and (16)). Consequently, banks are willing to supply more loans. This means that contrary to Equilibrium I, in Equilibrium II and III not only first stage marginal funding costs decrease but in addition, there is a reduction of expected second stage marginal funding costs. Therefore, the impact of monetary policy on bank loan supply is stronger than in Equilibrium I.

In Equilibrium II, the reinforcing effect increases in interbank market transaction costs. Crucial is the decrease of the expected liquidity deficit per unit of loans due to the increased borrowing from the refinancing operations, formally reflected by an increase in \bar{t}_i^{opt} . As the deficit will be costly refinanced in the interbank market, the impact of the decreased expected liquidity deficit on expected marginal funding costs is the stronger, the larger the interbank market transaction costs are. Furthermore, in both equilibria, II and III, the reinforcing effect is the stronger the higher the extent of uncertainty about a bank's net deposit transfer χ is. The reason is that the expected liquidity deficit per unit of loans is determined by this uncertainty (see equations (15) and (16)). Therefore, a large χ gives a

¹⁸Note that in Equilibrium II, the interbank rate has reached its lower bound so that an increase in the interbank rate is generally possible. This would however imply a switch from Equilibrium II to Equilibrium I. In this section, we thus only consider changes in i^{RO} which ensure that $\gamma > \frac{i^{RO} - i^{DF}}{2[1 - G(0)]}$, i.e. we stay in Equilibrium II and the interbank rate is fixed at its lower bound.

higher weight to the decrease of the expected deficit due to the increased borrowing from the refinancing operations.

Finally, Proposition 3 shows that the reinforcing effect is stronger in Equilibrium III than in Equilibrium II. The intuition behind this result is that covering a liquidity deficit by borrowing from the lending facility in Equilibrium III is more expensive than of using the interbank market in Equilibrium II.¹⁹

7.3 Main Policy Rate and the Rates on the Facilities

Alternatively, the central bank might change all its interest rates instead of solely changing its policy rate. In this case we obtain

Proposition 4: A likewise change in all central bank's interest rates, i.e. $di^{RO} = di^{DF} = di^{LF}$, implies for Equilibrium j

$$\frac{\partial L^{j*}}{\partial i^{RO}} = -\frac{c}{\lambda} < 0 \quad and \quad \frac{\partial^2 L^{j*}}{\partial i^{RO} \partial \chi} = \frac{\partial^2 L^{j*}}{\partial i^{RO} \partial \gamma} = 0 \quad \forall \quad j.$$

Proof: See appendix.

If the central bank changes all interest rates to the same extent, a reinforcing effect will be avoided. Recall from Proposition 3 that the reinforcing effect will occur if the interbank rate has already reached its lower bound so that the interbank rate cannot adjust to changes in the policy rate anymore (Equilibrium II) or if due to high transaction costs an interbank market does not exist (Equilibrium III). Changing the rates of the facilities leads to a change of the lower bound which allows the interbank rate to adjust in Equilibrium II as well. The price mechanism works again so that the quantity effect which is responsible for the reinforcing effect does not occur, i.e. $\bar{t}_i^{opt} = \bar{t}^*$ does not change. In Equilibrium III, a likewise change in the rates on the facilities as in the main policy rate implies that expected marginal revenues of borrowing from the refinancing operations change to the same extent as marginal costs. Consequently, banks do not change their borrowing behavior so that also in this equilibrium $\bar{t}_i^{opt} = \bar{t}^*$ does not change. A quantity effect, which is responsible for the reinforcing effect, does not occur. Therefore, in all equilibria a likewise change in all central bank's interest rates will only affect first stage

¹⁹Appendix A.3 reveals that relevant costs in Equilibrium III are given by $i^{LF} - i^{DF}$, in Equilibrium II they are 2γ . As in Equilibrium II $\gamma \leq \bar{\bar{\gamma}} = (i^{LF} - i^{DF})/2$, the relevant marginal funding costs in Equilibrium III are higher than in Equilibrium II.

²⁰Referring to the right hand side of (13), expected marginal revenues are given by $i^{DF}G(\bar{t}_i^{opt}) + i^{LF}[1 - G(\bar{t}_i^{opt})]$ in Equilibrium III.

marginal funding costs (see Lemma 2). This means that in all equilibria, this central bank behavior has the same impact on bank loan supply. A reinforcing effect does not occur.

However, these results point to the problem of a zero lower bound on interest rates. In Equilibrium II and III, the reinforcement of an expansionary monetary policy cannot be avoided by a likewise change in the rates on the facilities if the rate on the deposit facility i^{DF} is already equal to zero. As a result, the zero lower bound on the deposit facility rate may imply an unavoidable reinforcement effect of an expansionary monetary policy.

The findings of this subsection point to the importance of the rates on the facilities in the presence of uncertain liquidity needs and interbank market frictions. Also the next two subsections illustrate their high relevance. They show that in the presence of uncertain liquidity needs and sufficiently high interbank market frictions, the facilities represent an additional effective monetary policy instrument.

7.4 Changing the Width of the Interest Rate Corridor

If the central bank changes the width of the interest rate corridor around its policy rate, we will obtain

Proposition 5: Suppose, we have a symmetric interest rate corridor, i.e. $i^{RO} = \frac{i^{DF} + i^{LF}}{2}$. A change in the width of the interest rate corridor, i.e. $di^{LF} = -di^{DF}$ and $di^{RO} = 0$, implies

$$\begin{split} &\frac{\partial L^{\mathrm{I}*}}{\partial (i^{LF}-i^{DF})}=0,\\ &\frac{\partial L^{\mathrm{II}*}}{\partial (i^{LF}-i^{DF})}=-\frac{(1-c)\chi}{\lambda}\overline{t}^*<0,\\ &\frac{\partial L^{\mathrm{III}*}}{\partial (i^{LF}-i^{DF})}=-\frac{(1-c)\chi}{\lambda}E\left[t_i|t_i>\overline{t}_i^{opt}\right]\left[1-G\left(\overline{t}_i^{opt}\right)\right]<0, \end{split}$$

with

$$\begin{split} \frac{\partial^2 L^{\text{I}*}}{\partial (i^{LF}-i^{DF})\partial \chi} &= 0, \quad \frac{\partial^2 L^{\text{II}*}}{\partial (i^{LF}-i^{DF})\partial \chi} < 0, \quad \frac{\partial^2 L^{\text{III}*}}{\partial (i^{LF}-i^{DF})\partial \chi} < 0, \\ \frac{\partial^2 L^{\text{I}*}}{\partial (i^{LF}-i^{DF})\partial \gamma} &= 0, \quad \frac{\partial^2 L^{\text{II}*}}{\partial (i^{LF}-i^{DF})\partial \gamma} < 0, \quad \frac{\partial^2 L^{\text{III}*}}{\partial (i^{LF}-i^{DF})\partial \gamma} &= 0. \end{split}$$

Proof: See appendix.

Proposition 5 shows that by changing the width of the interest rate corridor, the central bank may influence banks' loan supply without changing its policy rate i^{RO} . The central bank thus has an additional instrument at hand.

Obviously, changing the width of the corridor has no effect in Equilibrium I. In this equilibrium, banks never use the facilities to balance their liquidity needs so that the rates on the facilities are irrelevant for their loan supply decision.

In Equilibrium II and III a bank may use one of the facilities. Consequently, the rates on the facilities become relevant for a bank's expected marginal funding costs, and therefore, for its optimal loan supply. The facilities are used to balance uncertain liquidity needs. If a bank has to use the deposit facility, from the ex post perspective it would have been better to borrow less from the refinancing operations as the bank pays for the liquidity the rate i^{RO} and it receives only the rate i^{DF} . Analogously, if a bank has to use the lending facility, from the expost perspective it would have been better to borrow more from the refinancing operations as then the bank would have paid for the liquidity only the rate i^{RO} instead of i^{LF} . Due to the uncertainty about actual liquidity needs additional funding costs may thus occur. These costs of uncertainty are the higher, the stronger the rates on the facilities deviate from the rate on the main refinancing operations. This means that an increase in the width of the corridor leads to an increase in expected marginal funding costs and corresponds therefore to a contractionary monetary policy. As in Equilibrium II expected marginal funding costs increase in γ and χ , the impact of this monetary policy impulse is the stronger the higher the frictions in the interbank market and the uncertainty about the net deposit transfer are. In Equilibrium III, in which the interbank market is not used, it is only the uncertainty about the net deposit transfer which reinforces the monetary policy impulse.

Note that if the zero lower bound on the deposit rate becomes binding, changing the width of the corridor will not be a feasible instrument for conducting contractionary monetary policy. However an expansionary monetary policy is still feasible.

7.5 Changing the Asymmetry of the Interest Rate Corridor

The central bank can also change the interest rate corridor in the sense that it becomes more or less asymmetric around the main policy rate. Then we obtain

Proposition 6: A change in the interest rate corridor in the form of $di^{LF} = di^{DF}$ and $di^{RO} = 0$, implies

$$\begin{split} &\frac{\partial L^{\text{I*}}}{\partial i^{DF}} = 0,\\ &\frac{\partial L^{\text{j*}}}{\partial i^{DF}} = \frac{(1-c)\chi}{2\lambda} \overline{t}^* > 0 \ \ \textit{for} \ \ j = \text{II, III,} \end{split}$$

with

$$\begin{split} &\frac{\partial^2 L^{\text{I}*}}{\partial i^{DF} \partial \chi} = 0 & \frac{\partial^2 L^{\text{II}*}}{\partial i^{DF} \partial \chi} > 0, & \frac{\partial^2 L^{\text{III}*}}{\partial i^{DF} \partial \chi} > 0, \\ &\frac{\partial^2 L^{\text{I}*}}{\partial i^{DF} \partial \gamma} = 0 & \frac{\partial^2 L^{\text{II}*}}{\partial i^{DF} \partial \gamma} > 0, & \frac{\partial^2 L^{\text{III}*}}{\partial i^{DF} \partial \gamma} = 0. \end{split}$$

Proof: See appendix.

As in Proposition 5, in Equilibrium I a change in the rates on the facilities has no impact on bank loan supply as the facilities will not be used and are not expected to be used. However, in the Equilibrium II and III, the central bank has an additional effective instrument at its disposal by changing the asymmetry of the interest rate corridor. If, for example, the central bank increases both rates but leaves its main rate unchanged, its monetary policy will be expansionary. In Equilibrium II, only the deposit facility may be used. Therefore, the increase in i^{DF} implies a decrease in the costs of uncertainty (see Subsection 7.4) leading to decreasing expected marginal funding costs. The increase in i^{LF} plays no role. In Equilibrium III, a bank may use the deposit facility or the lending facility. The higher i^{DF} implies lower costs of uncertainty resulting in lower expected marginal funding costs. The higher i^{LF} leads to higher costs of uncertainty, and therefore also to higher expected marginal funding costs. However, the former effect outweighs the latter: The increase in both, i^{DF} and i^{LF} , result in higher marginal revenues of borrowing from the refinancing operation (see Lemma 1). Consequently, banks actually start to borrow more from the refinancing operations and the probability of facing a liquidity surplus at the second stage increases. This means that the decreasing effect on expected marginal funding costs due to the increase in i^{DF} outweighs the increasing effect due to the increase in i^{LF} . As a result, this monetary policy impulse has a positive impact on bank loan supply. Analogously to the situation described in Section 7.2, in Equilibrium II, this monetary policy impulse is the more effective, the higher the transaction costs in the interbank market are, and in Equilibria II and III the impact of this monetary policy impulse on bank loan supply increases in the extent of uncertainty about the net deposit transfer.

7.6 Collateralization and Minimum Reserves

In our analysis, we have not considered the collateralization of central bank credits and minimum reserve requirements. Both are elements of the Eurosystem's operational framework. However, introducing these elements would not change the qualitative results of our analysis.

If we considered the collateralization of central bank credits, banks would face opportunity costs of holding collateral. As a result, expected marginal funding costs would increase.²¹ Bank loan supply would be lower but our model results would not change qualitatively.

Considering reserve requirements lead to a structural liquidity deficit of the banking sector. In our model, a structural liquidity deficit is already captured by considering cash withdrawals. Introducing reserve requirements would therefore simply increase the already existing structural deficit. A main feature of the Eurosystem's minimum reserve system is that banks can make use of averaging provision of required reserves during the reserve maintenance period. This allows banks to smooth out liquidity fluctuations. In our model, costs resulting from uncertain liquidity fluctuations, and thus banks' expected marginal funding costs would decrease. Although this would have a positive effect on bank loan supply, the qualitative results of our model would not change.

8 Summary

The interbank market is regarded to play a crucial role for the implementation of monetary policy as it serves as the starting point of the transmission mechanism. Based on a theoretical model, this paper analyzes in how far interbank market frictions in the form of transaction costs influence the effectiveness of monetary policy and draws conclusions for monetary policy implementation.

We show that independently of interbank market frictions monetary policy is effective. The central bank is able to influence banks' funding costs of granting loans, and therefore their loan supply, just by changing its main policy rate. However, frictions in the interbank market may reinforce this effect.

Generally, the central bank can steer this reinforcing effect by changing the rates on its facilities. This indicates that for sufficiently high interbank market transaction costs, the standing facilities present an additional effective monetary policy instrument. By changing the width or the asymmetry of the interest rate corridor the central bank can influence banks' expected marginal funding costs and therefore, their loan supply. It should be noted that lowering the rate on the deposit facility taken for itself, corresponds

²¹For a respective analysis see, for example, Neyer and Wiemers (2004) and Berentsen and Monnet (2008).

to a contractionary monetary policy. This implies that the zero lower bound of the rate on the deposit facility is not a problem as long as the central bank wants to conduct a an expansionary monetary policy.

Appendix

A.1 Proof of Lemma 1

Recall from (11) that the first stage optimization problem of bank i reads:

$$\max_{L_{i},RO_{i}} E\left[\pi_{i}\right] = i^{L} L_{i} - \frac{1}{2} \lambda L_{i}^{2} - i^{RO} RO_{i} - \max\left\{i^{IBM} - \gamma, i^{DF}\right\} \int_{t^{min}}^{\bar{t}_{i}} N_{i} g(t_{i}) dt_{i} \\
- \min\left\{i^{IBM} + \gamma, i^{LF}\right\} \int_{\bar{t}_{i}}^{t^{max}} N_{i} g(t_{i}) dt_{i}, \tag{21}$$

subject to (8) and (12). By applying the Leibniz rule and making use of the fact that $N_i = 0$ for $t_i = \bar{t}_i$, we obtain:

$$\frac{\partial E[\pi_i]}{\partial RO_i} = -i^{RO} - \max\left\{i^{IBM} - \gamma, i^{DF}\right\} \int_{t^{min}}^{\bar{t}_i} \frac{\partial N_i}{\partial RO_i} g(t_i) dt_i
- \min\left\{i^{IBM} + \gamma, i^{LF}\right\} \int_{\bar{t}_i}^{t^{max}} \frac{\partial N_i}{\partial RO_i} g(t_i) dt_i.$$
(22)

We can infer from (8) that $\frac{\partial N_i}{\partial RO_i} = -1$. Insertion of this in (22) and rewriting terms yields

$$\frac{\partial E[\pi_i]}{\partial RO_i} = -i^{RO} + \max\left\{i^{IBM} - \gamma, i^{DF}\right\}G\left(\bar{t}_i\right) + \min\left\{i^{IBM} + \gamma, i^{LF}\right\}\left[1 - G\left(\bar{t}_i\right)\right]. \tag{23}$$

Note that $\frac{\partial E[\pi_i]}{\partial RO_i}$ decreases in $G(\bar{t}_i) \in [0,1]$, which in turn (weakly) increases in \bar{t}_i . Moreover, we know from (12) that

- \bar{t}_i increases in RO_i , so that $\frac{\partial E[\pi_i]}{\partial RO_i}$ (weakly) decreases in RO_i ,
- and from the restriction $RO_i \ge 0$ that \bar{t}_i is restricted to $\bar{t}_i \ge -\frac{c}{(1-c)\chi} =: \tilde{t}$.

Denoting optima by the superscript opt, we can distinguish three cases:

- 1. If $i^{IBM} > i^{RO} + \gamma$, then $\frac{\partial E[\pi_i]}{\partial RO_i} > 0$ for all $G(\bar{t}_i)$. Therefore, we obtain $\bar{t}_i^{opt} = \infty$. In conjunction with (12), this yields $RO_i^{opt} = \infty$.
- 2. If $i^{IBM} \in \left[i^{RO} \gamma, i^{RO} + \gamma\right]$, then $\frac{\partial E[\pi_i]}{\partial RO_i} = 0$ only if $\bar{t}_i = \bar{t}_i^{opt}$, where \bar{t}_i^{opt} is implicitly defined by (13). In conjunction with (12) and the restriction $RO_i \geq 0$, this yields $RO_i^{opt} = \max\left\{0, \bar{t}_i^{opt} \left(1 c\right)\chi L_i^{opt} + cL_i^{opt}\right\}$, which brings us to two subcases.
 - If $\bar{t}_{i}^{opt} > \tilde{t}$ and thus $G\left(\bar{t}_{i}^{opt}\right) > G\left(\tilde{t}\right)$, then $RO_{i}^{opt} = \bar{t}_{i}^{opt}\left(1-c\right)\chi L_{i}^{opt} + cL_{i}^{opt} > 0$.
 - If $\bar{t}_{i}^{opt} \leq \tilde{t}$ and thus $G\left(\bar{t}_{i}^{opt}\right) \leq G\left(\tilde{t}\right)$, then $RO_{i}^{opt} = 0$.

3. If $i^{IBM} < i^{RO} - \gamma$, then $\frac{\partial E[\pi_i]}{\partial RO_i} < 0$ for all $G(\bar{t}_i)$. Therefore, we obtain $\bar{t}_i^{opt} = -\infty$. In conjunction with (12) and the restriction $RO_i \ge 0$, this yields $RO_i^{opt} = 0$.

Consequently, we have shown that only if $i^{IBM*} \in [i^{RO} - \gamma, i^{RO} + \gamma]$ a bank's optimal borrowing from the refinancing operation is described by the first order condition given by equation (13). Note that we restrict the analysis in the paper to $\bar{t}_i^{opt} \geq \tilde{t} := -\frac{c}{(1-c)\chi}$.

A.2 Proof of Lemma 2

From (12) and due to the restriction $RO_i \geq 0$ it follows that \bar{t}_i is restricted to $\bar{t}_i \geq -\frac{c}{(1-c)\chi} =: \tilde{t}$. By applying the Leibniz rule on (21) and making use of the facts that $N_i = 0$ for $t_i = \bar{t}_i$ and that optimal borrowing in the refinancing operation implies $\bar{t}_i = \max\left\{\bar{t}_i^{opt}, \tilde{t}\right\}$, we obtain:

$$\frac{\partial E[\pi_i]}{\partial L_i} = i^L - \lambda L_i - \max\left\{i^{IBM} - \gamma, i^{DF}\right\} \int_{t^{min}}^{\max\left\{\bar{t}_i^{opt}, \tilde{t}\right\}} \frac{\partial N_i}{\partial L_i} g(t_i) dt_i
- \min\left\{i^{IBM} + \gamma, i^{LF}\right\} \int_{\max\left\{\bar{t}_i^{opt}, \tilde{t}\right\}}^{t^{max}} \frac{\partial N_i}{\partial L_i} g(t_i) dt_i + \frac{\partial E[\pi_i]}{\partial RO_i^{opt}} \frac{\partial E[RO_i^{opt}]}{\partial L_i}.$$
(24)

We can infer from (8) and the envelope theorem that $\frac{\partial N_i}{\partial L_i} = c + (1-c)\chi t_i$ and $\frac{\partial E[\pi_i]}{\partial RO_i^{opt}} \frac{\partial E[RO_i^{opt}]}{\partial L_i} = 0$. Insertion of this in (24) and rewriting terms yields

$$\frac{\partial E[\pi_i]}{\partial L_i} = i^L - \lambda L_i - (1 - c) \chi \max \left\{ i^{IBM} - \gamma, i^{DF} \right\} \int_{t^{min}}^{\max \left\{ \bar{t}_i^{opt}, \tilde{t} \right\}} t_i g(t_i) dt_i
- (1 - c) \chi \min \left\{ i^{IBM} + \gamma, i^{LF} \right\} \int_{\max \left\{ \bar{t}_i^{opt}, \tilde{t} \right\}}^{t^{max}} t_i g(t_i) dt_i
- c \max \left\{ i^{IBM} - \gamma, i^{DF} \right\} G\left(\max \left\{ \bar{t}_i^{opt}, \tilde{t} \right\} \right)
- c \min \left\{ i^{IBM} + \gamma, i^{LF} \right\} \left[1 - G\left(\max \left\{ \bar{t}_i^{opt}, \tilde{t} \right\} \right) \right].$$
(25)

This brings us to two cases.

• If $\bar{t}_i^{opt} > \tilde{t}$ and thus $G\left(\bar{t}_i^{opt}\right) > G\left(\tilde{t}\right)$, then insertion of (13) in (25) implies that $\frac{\partial E[\pi_i]}{\partial L_i} = 0$ only if (14) is met.

• If $\bar{t}_{i}^{opt} \leq \tilde{t}$ and thus $G\left(\bar{t}_{i}^{opt}\right) \leq G\left(\tilde{t}\right)$, then $\frac{\partial E[\pi_{i}]}{\partial L_{i}} = 0$ only if

$$i^{L} = \lambda L_{i}^{opt} + ci^{DF}G\left(\tilde{t}\right) + ci^{LF}\left[1 - G\left(\tilde{t}\right)\right]$$

$$+ (1 - c)\chi \max\left\{i^{IBM} - \gamma, i^{DF}\right\} E\left[t_{i}|t_{i} < \tilde{t}\right] G\left(\tilde{t}\right)$$

$$+ (1 - c)\chi \min\left\{i^{IBM} + \gamma, i^{LF}\right\} E\left[t_{i}|t_{i} > \tilde{t}\right] \left[1 - G\left(\tilde{t}\right)\right].$$
(26)

Note that we restrict the analysis in the paper to $\bar{t}_i^{opt} \geq \tilde{t} := -\frac{c}{(1-c)\gamma}$

A.3 A Bank's Expected Second Stage Marginal Funding Costs

Expected second-stage marginal funding costs in the case of a liquidity surplus are given by

$$(1-c)\chi \max\left\{i^{IBM} - \gamma, i^{DF}\right\} \frac{\int_{t_{min}}^{\overline{t_{i}^{opt}}} t_{i}g(t_{i}) dt_{i}}{G(\overline{t_{i}^{opt}})},$$

while expected second stage-marginal funding costs in the case of a liquidity deficit are

$$(1-c)\chi \min\left\{i^{IBM} + \gamma, i^{LF}\right\} \frac{\int_{\overline{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i}{1 - G(\overline{t}_i^{opt})}.$$

As the former occurs with the probability $G(\bar{t}_i^{opt})$ and the latter with the probability $1 - G(\bar{t}_i^{opt})$, expected second-stage marginal funding costs are given by

$$(1 - c)\chi \max \left\{i^{IBM} - \gamma, i^{DF}\right\} E\left[t_i | t_i < \overline{t}_i^{opt}\right] G\left(\overline{t}_i^{opt}\right) + (1 - c)\chi \min \left\{i^{IBM} + \gamma, i^{LF}\right\} E\left[t_i | t_i > \overline{t}_i^{opt}\right] \left[1 - G\left(\overline{t}_i^{opt}\right)\right].$$
(27)

Active Interbank Market $(\gamma \leq \bar{\gamma})$

Equation (15) reveals that these costs are formally the same for all cases described in Proposition 1 in which the interbank market is active. For interpreting this term in more detail, it is useful to consider that, due to E[T] = 0, the expected second stage liquidity deficit per unit of loans equals the negative value of the expected second stage liquidity surplus per unit of loans:

$$(1-c)\chi \int_{\bar{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i = -(1-c)\chi \int_{t^{min}}^{\bar{t}_i^{opt}} t_i g(t_i) dt_i.$$
 (28)

If RO = cL, banks will balance their liquidity needs solely via the interbank market and the facilities will not be used. Considering (14), expected second stage marginal funding costs are then

$$(1 - c)\chi \int_{t^{min}}^{\bar{t}_i^{opt}} t_i g(t_i) dt_i (i^{IBM*} - \gamma) + (1 - c)\chi \int_{\bar{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i (i^{IBM*} + \gamma).$$
 (29)

The interbank rate constitutes negative marginal funding costs in the case of an individual liquidity surplus and positive marginal funding costs in the case of an individual liquidity deficit, while transaction costs have a positive impact on marginal funding costs in both cases. Expression (29) shows that the effects of the interbank rate on expected marginal funding costs compensate each other so that only interbank market transaction costs are relevant for a bank's expected second stage marginal funding costs. Considering (28), the expected share per unit of loans for which funding costs are expected to accrue is $2(1-c)\chi \int_{\overline{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i = 2(1-c)\chi E[t_i|t_i > \overline{t}_i^{opt}][1-G(\overline{t}_i^{opt})]$, the relevant funding costs are γ , so that (29) is equivalent to (15).

If RO > cL, banks will cover their individual liquidity deficit in the interbank market at marginal costs of $i^{IBM*} + \gamma = i^{DF} + 2\gamma$. In the case of an individual liquidity surplus, they place their excess liquidity in the interbank market or in the deposit facility so that marginal revenues are given by i^{DF} . Consequently, expected second stage marginal funding costs are

$$(1 - c)\chi \int_{t^{min}}^{\bar{t}_i^{opt}} t_i g(t_i) dt_i i^{DF} + (1 - c)\chi \int_{\bar{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i (i^{DF} + 2\gamma), \tag{30}$$

which is again equivalent to (15). The expected marginal funding costs given by (30) show that the interest rate i^{DF} has a negative impact on these costs in the case of a liquidity surplus and a positive impact in the case of a liquidity deficit. As these effects compensate each other, the expected share per unit of loans for which those funding costs are expected to accrue is equal to the expected liquidity deficit per unit of loans given by $(1-c)\chi \int_{\bar{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i = (1-c)\chi E[t_i|t_i > \bar{t}_i^{opt}][1-G(\bar{t}_i^{opt})]$, the relevant funding costs are 2γ . Transaction costs are relevant for two reasons. First, they will accrue if the bank borrows liquidity in the interbank market and, second, because they imply a higher interbank rate which is $i^{DF} + \gamma$.

Analogously, if RO < cL, the expected share per unit of loans for which funding costs are expected to accrue is $-(1-c)\chi \int_{t^{min}}^{\bar{t}_i^{opt}} t_i g(t_i) dt_i = (1-c)\chi \int_{\bar{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i = (1-c)\chi E[t_i|t_i > \bar{t}_i^{opt}][1-G(\bar{t}_i^{opt})]$, and the relevant funding costs are 2γ .

Inactive Interbank Market $(\gamma > \bar{\gamma})$

In case there is no interbank market as $\gamma > \bar{\bar{\gamma}}$, expected second stage marginal funding costs are

$$(1 - c)\chi \int_{t^{min}}^{\bar{t}_i^{opt}} t_i g(t_i) dt_i i^{DF} + (1 - c)\chi \int_{\bar{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i i^{LF}.$$
 (31)

Considering (28) we obtain that expected second stage marginal funding costs are equal to (16). The expected share per unit of loans for which funding costs are expected to accrue is $(1-c)\chi \int_{\overline{t}_i^{opt}}^{t^{max}} t_i g(t_i) dt_i = (1-c)\chi E[t_i|t_i > \overline{t}_i^{opt}][1-G(\overline{t}_i^{opt})]$, and the relevant funding costs are $i^{LF} - i^{DF}$.

A.4 Proof of Proposition 2

Following Proposition 1, we distinguish between an active and an inactive interbank market to determine the interbank rate and bank aggregate borrowing and lending in equilibrium.

Active Interbank Market $(\gamma \leq \bar{\gamma})$

It is useful to distinguish between the three cases described in Proposition 1:

1. Suppose that an equilibrium exists with $RO^* < cL^*$. Then, (10) implies $i^{IBM^*} = i^{LF} - \gamma$ while according to (12) $RO^* = \overline{t}^* (1-c) \chi L^* + cL^* \geq 0$ implies $\overline{t} < 0$ and thus

$$G(\bar{t}) < G(0)$$
.

Insertion of i^{IBM^*} in (13) yields

$$G\left(\overline{t}\right) = \frac{i^{LF} - i^{RO}}{2\gamma} < G\left(0\right).$$

For all $\gamma \in [0, \bar{\gamma}]$ it follows due to $i^{LF} - i^{RO} \ge i^{RO} - i^{DF}$ that $\frac{i^{LF} - i^{RO}}{2\gamma} > 0.5$. As we assume that G(0) < 0.5, $RO^* < cL^*$ does not constitute an equilibrium.

2. Suppose that an equilibrium exists with $RO^* = cL^*$. Then, (10) implies

$$i^{IBM^*} \in \left[i^{DF} + \gamma, i^{LF} - \gamma\right],$$

while according to (12) $RO^* = \overline{t}^* (1-c) \chi L^* + cL^* > 0$ implies $\overline{t} = 0$ and thus $G(\overline{t}) = G(0)$. Insertion of $G(\overline{t})$ in (13) yields

$$i^{IBM^{*}}=i^{RO}-\gamma+2\gamma G\left(0\right) ,$$

and thus $G(0) \in \left[\frac{i^{DF} + 2\gamma - i^{RO}}{2\gamma}, \frac{i^{LF} - i^{RO}}{2\gamma}\right]$. As $RO^* = cL^*$, there is no aggregate liquidity deficit at the second stage so that neither the lending nor the deposit facility is used.

3. Suppose that an equilibrium exists with $RO^* > cL^*$. Then, (10) implies $i^{IBM^*} = i^{DF} + \gamma$ while according to (12) $RO^* = \bar{t}^* (1-c) \chi L^* + cL^* > 0$ implies $\bar{t} > 0$ and thus

$$G\left(\overline{t}\right) > G\left(0\right)$$
.

Insertion of i^{IBM^*} in (13) yields

$$G\left(\bar{t}\right) = \frac{i^{DF} + 2\gamma - i^{RO}}{2\gamma} > G\left(0\right).$$

As $RO^* > cL^*$, banks have to place their aggregate liquidity surplus of the second stage in the deposit facility so that $DF^* > 0$.

Inactive Interbank Market $(\gamma > \bar{\gamma})$

It is useful to distinguish between the same three cases as for an active interbank market:

1. Suppose that an equilibrium exists with $RO^* < cL^*$. Then, we have $i^{IBM^*} \in [i^{LF} - \gamma, i^{DF} + \gamma]$ while $RO^* = \bar{t}^* (1 - c) \chi L^* + cL^* > 0$ implies $\bar{t} < 0$ and thus

$$G\left(\overline{t}\right) < G\left(0\right)$$
.

Insertion of i^{IBM^*} in (13) yields

$$G\left(\overline{t}\right) = \frac{i^{LF} - i^{RO}}{i^{LF} - i^{DF}} < G\left(0\right).$$

It follows due to $i^{LF} - i^{RO} \ge i^{RO} - i^{DF}$ that $\frac{i^{LF} - i^{RO}}{i^{LF} - i^{DF}} > 0.5$. As we assume that G(0) < 0.5, $RO^* < cL^*$ does not constitute an equilibrium.

2. Suppose that an equilibrium exists with $RO^* = cL^*$. Then, we have $i^{IBM^*} \in [i^{LF} - \gamma, i^{DF} + \gamma]$ while $RO^* = \bar{t}^* (1-c) \chi L^* + cL^* > 0$ implies $\bar{t} = 0$ and thus

$$G\left(\overline{t}\right) = G\left(0\right).$$

Insertion of i^{IBM^*} in (13) yields again

$$G\left(\overline{t}\right) = \frac{i^{LF} - i^{RO}}{i^{LF} - i^{DF}} = G\left(0\right).$$

Due to G(0) < 0.5, $RO^* = cL^*$ does not constitute an equilibrium either.

3. Suppose that an equilibrium exists with $RO^* > cL^*$. Then, we have $i^{IBM^*} \in [i^{LF} - \gamma, i^{DF} + \gamma]$ while $RO^* = \overline{t}^* (1 - c) \chi L^* + cL^* > 0$ implies $\overline{t} > 0$ and thus

$$G(\bar{t}) > G(0)$$
.

Insertion of i^{IBM^*} in (13) yields again

$$G\left(\bar{t}\right) = \frac{i^{LF} - i^{RO}}{i^{LF} - i^{DF}} > G\left(0\right).$$

Banks with a liquidity deficit have to borrow from the lending facility, while banks with a liquidity surplus have to place their excess liquidity in the deposit facility. As $RO^* > cL^*$, it follows that $DF^* > LF^* > 0$.

A.5 Impact of Uncertainty and Transaction Costs on Loan Supply

In this subsection, we present the derivation of the argument presented in Section 7 that an increase in both uncertainty and frictions in the interbank market have a negative effect on bank loan supply.

It follows from (14) and Proposition 2 that

$$L^{j*} = \frac{1}{\lambda} \left[i^L - ci^{RO} - \left[(1 - c)\chi \int_{\overline{t}^*}^{t^{max}} t_i g(t_i) dt_i \right] \xi \right]$$
 (32)

with

$$\xi = \begin{cases} 2\gamma & \text{if } \gamma \leq \bar{\bar{\gamma}}, \\ i^{LF} - i^{DF} & \text{if } \gamma > \bar{\bar{\gamma}}. \end{cases}$$
 (33)

Note that $\bar{t}_i^* = \bar{t}_i^{opt}$ is only determined by the interest rates and γ . It follows from (12) that \bar{t}_i will change c.p. if L_i changes. However, \bar{t}_i^{opt} will not change, as a change in L_i induces a respective change in RO_i to fulfill the optimality condition given by (13). Consequently, (32) represents the reduced form of L^{j*} which is an important aspect for our comparative static analysis.

Uncertainty

Applying the Leibniz rule on (32), the first derivative w.r.t. χ reads

$$\frac{\partial L^{j*}}{\partial \chi} = -\frac{(1-c)\xi}{\lambda} \left[\int_{\overline{t}^*}^{t^{max}} t_i g(t_i) dt_i - \chi \frac{\partial \overline{t}^*}{\partial \chi} \overline{t}^* g(\overline{t}^*) \right].$$

We derive from (13) the function

$$F := \max\left\{i^{IBM} - \gamma, i^{DF}\right\}G\left(\overline{t}^{*}\right) + \min\left\{i^{IBM} + \gamma, i^{LF}\right\}\left[1 - G\left(\overline{t}^{*}\right)\right] - i^{RO} = 0. \quad (34)$$

Applying the implicit function theorem on (34) thus yields

$$\frac{\partial \bar{t}^*}{\partial \chi} = -\frac{\frac{\partial F}{\partial \chi}}{\frac{\partial F}{\partial \bar{t}^*}} = 0 \quad \forall \quad \bar{t}^*$$
 (35)

so that

$$\frac{\partial L^{j*}}{\partial \chi} = -\frac{(1-c)\xi}{\lambda} E\left[t_i | t_i > \overline{t}_i^*\right] \left[1 - G\left(\overline{t}_i^*\right)\right] < 0 \ \forall \ j.$$

Transaction Costs

If $\gamma \leq \bar{\bar{\gamma}}$, it follows from (33) that $\xi = 2\gamma$. Applying the Leibniz rule on (32), the first derivative w.r.t. γ then reads

$$\frac{\partial L^{j*}}{\partial \gamma} = -\frac{(1-c)2\chi}{\lambda} \left[\int_{\overline{t}^*}^{t^{max}} t_i g(t_i) dt_i - \gamma \frac{\partial \overline{t}^*}{\partial \gamma} \overline{t}^* g(\overline{t}^*) \right].$$

Applying the implicit function theorem on (34) yields

$$\frac{\partial \overline{t}^*}{\partial \gamma} = -\frac{\frac{\partial F}{\partial \gamma}}{\frac{\partial F}{\partial \overline{t}^*}} = -\frac{G(\overline{t}^*)}{\gamma g(\overline{t}^*)} \qquad \text{if } j = I,$$
(36)

$$\frac{\partial \overline{t}^*}{\partial \gamma} = -\frac{\frac{\partial F}{\partial \gamma}}{\frac{\partial F}{\partial \overline{t}^*}} = \frac{1 - G(\overline{t}^*)}{\gamma g(\overline{t}^*)} \qquad \text{if } j = \text{II},$$
 (37)

so that

$$\begin{split} \frac{\partial L^{\mathrm{I}*}}{\partial \gamma} &= \frac{2(1-c)\chi}{\lambda} \left[E\left[t_i | t_i < \overline{t}^*\right] - \overline{t}^*\right] G\left(\overline{t}^*\right) < 0, \\ \frac{\partial L^{\mathrm{II}*}}{\partial \gamma} &= -\frac{2(1-c)\chi}{\lambda} \left[E\left[t_i | t_i \geq \overline{t}^*\right] - \overline{t}^*\right] \left[1 - G(\overline{t}^*)\right] < 0. \end{split}$$

If $\gamma > \bar{\gamma}$, it follows from (33) that $\xi = i^{LF} - i^{DF}$. Applying the implicit function theorem on (34) yields

$$\frac{\partial \bar{t}^*}{\partial \gamma} = -\frac{\frac{\partial F}{\partial \gamma}}{\frac{\partial F}{\partial \bar{t}^*}} = 0 \tag{38}$$

so that $\frac{\partial L^{\text{III}*}}{\partial \gamma} = 0$.

Moreover, it follows from (35) that the mixed partial derivative with respect to χ reads

$$\frac{\partial^{2} L^{\text{I*}}}{\partial \gamma \partial \chi} = \frac{2(1-c)}{\lambda} \left[E\left[t_{i} | t_{i} < \overline{t}^{*}\right] - \overline{t}^{*}\right] G\left(\overline{t}^{*}\right) < 0,$$

$$\frac{\partial^{2} L^{\text{II*}}}{\partial \gamma \partial \chi} = -\frac{2(1-c)}{\lambda} \left[E\left[t_{i} | t_{i} \geq \overline{t}^{*}\right] - \overline{t}^{*}\right] \left[1 - G(\overline{t}^{*})\right] < 0.$$

A.6 Proof of Proposition 3

We proof this proposition in two steps. First, we determine the derivative with respect to i^{RO} for each feasible equilibrium. Afterwards, we derive the respective mixed partial derivative with respect to χ and γ .

1. Applying the Leibniz rule on (32), the first derivative w.r.t. i^{RO} reads

$$\frac{\partial L^{j*}}{\partial i^{RO}} = -\frac{1}{\lambda} \left[c - (1-c)\chi \frac{\partial \overline{t}^*}{\partial i^{RO}} \overline{t}^* g(\overline{t}^*) \xi \right] \quad \forall \quad j.$$

Applying the implicit function theorem on (34) yields

$$\frac{\partial \overline{t}^*}{\partial i^{RO}} = -\frac{\frac{\partial F}{\partial i^{RO}}}{\frac{\partial F}{\partial \overline{t}^*}} = 0 \qquad \text{if } j = I,$$

$$\frac{\partial \overline{t}^*}{\partial i^{RO}} = -\frac{\frac{\partial F}{\partial i^{RO}}}{\frac{\partial F}{\partial \overline{t}^*}} = -\frac{1}{\xi g(\overline{t}^*)} \qquad \text{if } j = \{II, III\}.$$

As $\overline{t}^* = 0$ for j = I, it follows for all j

$$\frac{\partial L^{j*}}{\partial i^{RO}} = -\frac{1}{\lambda} \left[c + (1 - c)\chi \bar{t}^* \right].$$

If $\gamma \leq \bar{\gamma}$, it follows that $2\gamma < i^{LF} - i^{DF}$. In Equilibrium II expected marginal revenues of borrowing from the refinancing operations are given by

$$i^{DF} + 2\gamma \left[1 - G\left(\overline{t}_i^{opt}\right) \right],$$
 (39)

while in Equilibrium III they read

$$i^{DF}G\left(\bar{t}_{i}^{opt}\right) + i^{LF}\left[1 - G\left(\bar{t}_{i}^{opt}\right)\right] \tag{40}$$

Comparing (39) and (40) shows that $G(\overline{t}_i^{opt})^{\text{II}} < G(\overline{t}_i^{opt})^{\text{III}}$ so that $\overline{t}_i^{opt} < \overline{t}_i^{opt}$. Due to $L_i^{opt} = L^*$ and $RO_i^{opt} = RO^*$, it follows that $\overline{t}^{*\text{II}} < \overline{t}^{*\text{III}}$ and thus $\frac{\partial L^{\text{III}*}}{\partial i^{RO}} \leq \frac{\partial L^{\text{II}*}}{\partial i^{RO}} < \frac{\partial L^{\text{II}*}}{\partial i^{RO}} < 0$.

2. (a) In order to determine the mixed partial derivative with respect to χ , we make use of the result obtained in (35). It thus follows that

$$\frac{\partial^2 L^{j*}}{\partial i^{RO} \partial \chi} = -\frac{1-c}{\lambda} \overline{t}^*,$$

so that $\frac{\partial^2 L^{\text{III}*}}{\partial i^{RO} \partial \chi} < \frac{\partial^2 L^{\text{II}*}}{\partial i^{RO} \partial \chi} < 0$ and $\frac{\partial^2 L^{\text{I}*}}{\partial i^{RO} \partial \chi} = 0$.

(b) In order to determine the mixed partial derivative with respect to γ , we make use of the results obtained in (36), (37) and (38). It thus follows that

$$\begin{split} \frac{\partial^2 L^{\text{I}*}}{\partial i^{RO} \partial \gamma} &= & 0, \\ \frac{\partial^2 L^{\text{II}*}}{\partial i^{RO} \partial \gamma} &= & -\frac{(1-c)\,\chi}{\lambda} \frac{\partial \overline{t}^*}{\partial \gamma} &= -\frac{(1-c)\,\chi \left[1-G(\overline{t}^*)\right]}{\lambda \gamma g(\overline{t}^*)} < 0, \\ \frac{\partial^2 L^{\text{III}*}}{\partial i^{RO} \partial \gamma} &= & 0. \end{split}$$

A.7 Proof of Proposition 4

We proof this proposition in two steps. First, we apply the total derivative to determine the impact of a change in the overall interest rate level. Afterwards, we derive the respective mixed partial derivative with respect to χ and γ .

1. Given $di^{LF} = di^{RO} = di^{DF}$, applying the total derivative on (32) yields

$$dL^{j*} = \frac{1}{\lambda} \left[-cdi^{RO} - \left[(1-c)\chi d\overline{t}^* \frac{\partial}{\partial \overline{t}^*} \int_{\overline{t}^*}^{t^{max}} t_i g(t_i) dt_i \right] \xi \right].$$

Moreover, applying the total derivative on (34) yields

$$\begin{split} \frac{\partial G\left(\overline{t}^*\right)}{\partial \overline{t}^*} d\overline{t}^* &= 0 \quad \text{if} \quad j = \text{I}, \\ di^{DF} - 2\gamma \frac{\partial G\left(\overline{t}^*\right)}{\partial \overline{t}^*} d\overline{t}^* - di^{RO} &= 0 \quad \text{if} \quad j = \text{II}, \\ di^{LF} - \left(i^{DF} - i^{LF}\right) \frac{\partial G\left(\overline{t}^*\right)}{\partial \overline{t}^*} d\overline{t}^* + \left(di^{DF} - di^{LF}\right) G\left(\overline{t}^*\right) - di^{RO} &= 0 \quad \text{if} \quad j = \text{III}. \end{split}$$

Due to $di^{RO}=di^{DF}=di^{LF}$ it follows for all j that $d\bar{t}^*=0$ so that

$$\frac{dL^{j*}}{di^{RO}} = -\frac{c}{\lambda}. (41)$$

2. It follows directly from (41) that $\frac{\partial^2 L^{j*}}{\partial i^{RO} \partial \chi} = \frac{\partial^2 L^{j*}}{\partial i^{RO} \partial \gamma} = 0$ for all j.

A.8 Proof of Proposition 5

We proof this proposition analogously to the proof of Proposition 4 in two steps. First, we apply the total derivative to determine the impact of a change in the rates of the facilities. Afterwards, we derive the respective mixed partial derivative with respect to χ and γ .

1. Given $di^{LF} = -di^{DF}$ and $di^{RO} = 0$, applying the total derivative on (32) yields

$$\begin{split} dL^{j*} &= -\frac{2\gamma(1-c)}{\lambda}\chi d\overline{t}^* \frac{\partial}{\partial \overline{t}^*} \int_{\overline{t}^*}^{t^{max}} t_i g(t_i) dt_i & \text{if } j = \{\text{I}, \text{II}\}, \\ dL^{j*} &= -\frac{1-c}{\lambda}\chi \left[(i^{LF} - i^{DF}) d\overline{t}^* \frac{\partial}{\partial \overline{t}^*} \int_{\overline{t}^*}^{t^{max}} t_i g(t_i) dt_i \\ &+ 2 di^{LF} \int_{\overline{t}^*}^{t^{max}} t_i g(t_i) dt_i \right] & \text{if } j = \text{III}. \end{split}$$

Moreover, applying the total derivative on (34) yields

$$\begin{split} \frac{\partial G\left(\overline{t}^{*}\right)}{\partial\overline{t}^{*}}d\overline{t}^{*} &= 0 & \text{if} \quad j = \mathrm{I}, \\ di^{DF} - 2\gamma \frac{\partial G\left(\overline{t}^{*}\right)}{\partial\overline{t}^{*}}d\overline{t}^{*} &= 0 & \text{if} \quad j = \mathrm{II}. \end{split}$$

As long as the interest rate corridor is symmetric, it follows for $\gamma > \bar{\gamma}$ that $G(\bar{t}^*) =$ 0.5 so that

$$d\bar{t}^* = 0$$
 if $i = III$.

Considering $\frac{\partial i^{LF}}{\partial (i^{LF}-i^{DF})}=0.5$ and $\frac{\partial i^{DF}}{\partial (i^{LF}-i^{DF})}=-0.5$ it follows

$$\frac{\partial L^{\text{I*}}}{\partial (i^{LF} - i^{DF})} = \frac{dL^{\text{I*}}}{di^{LF}} \frac{\partial i^{LF}}{\partial (i^{LF} - i^{DF})} = 0, \tag{42}$$

$$\frac{\partial L^{\text{I*}}}{\partial (i^{LF} - i^{DF})} = \frac{dL^{\text{I*}}}{di^{LF}} \frac{\partial i^{LF}}{\partial (i^{LF} - i^{DF})} = 0,$$

$$\frac{\partial L^{\text{II*}}}{\partial (i^{LF} - i^{DF})} = \frac{dL^{\text{II*}}}{di^{DF}} \frac{\partial i^{DF}}{\partial (i^{LF} - i^{DF})} = -\frac{(1 - c)\chi}{\lambda} \bar{t}^* < 0,$$
(42)

$$\frac{\partial L^{\text{III}*}}{\partial (i^{LF} - i^{DF})} = \frac{dL^{\text{III}*}}{di^{LF}} \frac{\partial i^{LF}}{\partial (i^{LF} - i^{DF})} = -\frac{(1 - c)\chi}{\lambda} E\left[t_i | t_i > \overline{t}_i^*\right] \left[1 - G\left(\overline{t}_i^*\right)\right] < 0.$$
(44)

2. Making use of the result obtained in (35), it follows directly from (42) to (44) that

$$\begin{split} &\frac{\partial^2 L^{\text{I}*}}{\partial (i^{LF}-i^{DF})\partial \chi}=0,\\ &\frac{\partial^2 L^{\text{II}*}}{\partial (i^{LF}-i^{DF})\partial \chi}=-\frac{(1-c)}{\lambda}\overline{t}^*<0,\\ &\frac{\partial^2 L^{\text{III}*}}{\partial (i^{LF}-i^{DF})\partial \chi}=-\frac{(1-c)}{\lambda}E\left[t_i|t_i>\overline{t}_i^*\right]\left[1-G\left(\overline{t}_i^*\right)\right]<0. \end{split}$$

Making use of the result obtained in (37) and (38), it follows that

$$\begin{split} &\frac{\partial^2 L^{\mathrm{j}*}}{\partial (i^{LF}-i^{DF})\partial \gamma} = 0 \ \text{ for } \ j = \mathrm{II, \, III,} \\ &\frac{\partial^2 L^{\mathrm{II}*}}{\partial (i^{LF}-i^{DF})\partial \gamma} = -\frac{(1-c)}{\lambda}\frac{1-G(\overline{t}^*)}{\gamma g(\overline{t}^*)} < 0. \end{split}$$

A.9 Proof of Proposition 6

We proof this proposition in the same two steps as in the proof of Proposition 5.

1. Given $di^{LF} = di^{DF}$ and $di^{RO} = 0$, applying the total derivative on (32) yields

$$dL^{j*} = -\frac{2\gamma(1-c)}{\lambda}\chi d\bar{t}^* \frac{\partial}{\partial \bar{t}^*} \int_{\bar{t}^*}^{t^{max}} t_i g(t_i) dt_i \qquad \text{if } j = \{\text{I}, \text{II}\},$$

$$dL^{j*} = -\frac{1-c}{\lambda}\chi(i^{LF} - i^{DF}) d\bar{t}^* \frac{\partial}{\partial \bar{t}^*} \int_{\bar{t}^*}^{t^{max}} t_i g(t_i) dt_i \qquad \text{if } j = \text{III}.$$

Moreover, applying the total derivative on (34) yields

$$\begin{split} \frac{\partial G\left(\overline{t}^*\right)}{\partial \overline{t}^*} d\overline{t}^* &= 0 \quad \text{if} \quad j = \mathrm{I}, \\ di^{DF} - 2\gamma \frac{\partial G\left(\overline{t}^*\right)}{\partial \overline{t}^*} d\overline{t}^* &= 0 \quad \text{if} \quad j = \mathrm{II}, \\ di^{LF} - (i^{LF} - i^{DF}) \frac{\partial G\left(\overline{t}^*\right)}{\partial \overline{t}^*} d\overline{t}^* &= 0 \quad \text{if} \quad j = \mathrm{III}, \end{split}$$

so that

$$\frac{\partial L^{I*}}{\partial i^{DF}} = 0, (45)$$

$$\frac{\partial L^{j*}}{\partial i^{DF}} = \frac{(1-c)\chi}{\lambda} \bar{t}^* > 0 \text{ for } j = \text{II, III.}$$
(46)

2. Making use of the result obtained in (35), it follows directly from (45) to (46) that

$$\begin{split} &\frac{\partial^2 L^{\text{I*}}}{\partial i^{DF} \partial \chi} = 0, \\ &\frac{\partial^2 L^{\text{j*}}}{\partial i^{DF} \partial \chi} = \frac{(1-c)}{\lambda} \overline{t}^* > 0 \ \text{ for } \ j = \text{II, III.} \end{split}$$

Making use of the result obtained in (37) and (38), it follows that

$$\begin{split} &\frac{\partial^2 L^{j*}}{\partial i^{DF} \partial \gamma} = 0 \quad \text{for} \quad j = \text{I, III,} \\ &\frac{\partial^2 L^{\text{II*}}}{\partial i^{DF} \partial \gamma} = \frac{(1-c)}{\lambda} \frac{1-G(\overline{t}^*)}{\gamma g(\overline{t}^*)} > 0. \end{split}$$

Bibliography

ALLEN, F., E. CARLETTI, AND D. GALE (2009): "Interbank Market Liquidity and Central Bank Intervention," *Journal of Monetary Economics*, 56(5), 639–652.

ANGELONI, I., A. K. KASHYAP, B. MOJON, AND D. TERLIZZESE (2003): "Monetary Transmission in the Euro Area: Does the Interest Rate Channel Explain it All?," NBER Working Paper No. 9984, National Bureau of Economic Research.

- ASHCRAFT, A., AND D. DUFFIE (2007): "Systemic Illiquidity in the Federal Funds Market," American Economic Review, 97(2), 221–225.
- Ayuso, J., and R. Repullo (2001): "Why Did the Banks Overbid? An Empirical Model of the Fixed Rate Tenders of the European Central Bank," *Journal of International Money and Finance*, 20(6), 857–870.
- Bartolini, L., G. Bertola, and A. Prati (2001): "Banks' Reserve Management, Transaction Costs, and the Timing of Federal Reserve Intervention," *Journal of Banking and Finance*, 25(7), 1287–1317.
- ———— (2002): "Day-To-Day Monetary Policy and the Volatility of the Federal Funds Interest Rate," Journal of Money, Credit, and Banking, 34(1), 137–159.
- BEAUPAIN, R., AND A. DURRÉ (2008): "The Interday and Intraday Patterns of the Overnight Market Evidence from an Electronic Platform," European Central Bank Working Paper No. 988.
- Berentsen, A., and C. Monnet (2008): "Monetary Policy in a Channel System," Journal of Monetary Economics, 55(6), 1067–1080.
- Bernanke, B. S., and M. Gertler (1989): "Agency Costs, Net Worth, and Business Fluctuations," *American Economic Review*, 79, 14–33.
- Bernanke, B. S., M. Gertler, and S. Gilchrist (1999): "The Financial Accelerator in a Quantitative Business Cycle Framework," in *Handbook of Macroeconomics*, ed. by J. B. Taylor, and M. Woodford, pp. 1341–1393, Amsterdam. Elsevier.
- BINDSEIL, U. (2005): "Over- and Underbidding in Central Bank Open Market Operations Conducted as Fixed Rate Tender," *German Economic Review*, 6(1), 95–130.
- BINDSEIL, U., AND P. J. KÖNIG (2011): "The Economics of TARGET2 Balances," SFB 649 Discussion Paper No. 2011-035, Humboldt-University Berlin.
- BOIVIN, J., M. T. KILEY, AND F. S. MISHKIN (2010): "How Has the Monetary Transmission Mechanism Evolved Over Time?," in *Handbook of Monetary Economics*, ed. by B. M. Friedman, and M. Woodford, vol. 3a, pp. 369–422, Amsterdam. Elsevier.
- BORIO, C., AND P. DISYATAT (2009): "Unconventional Monetary Policy: An Appraisal," BIS Working Papers No. 292, Bank for International Settlements.
- Bräuning, F., and F. Fecht (2012): "Relationship Lending in the Interbank Market and the Price of Liquidity," Deutsche Bundesbank Discussion Paper No. 22/2012.
- CASSOLA, N., AND M. HUETL (2010): "The Euro Overnight Interbank Market and ECB's Liquidity Management Policy During Tranquil and Turbulent Times," Paper presented at the ECB Workshop on "Challenges to Monetary Policy Implementation beyond the Financial Market Turbulence", in Frankfurt/Main, Germany on 1 December 2009. Available at http://www.ecb.de/events/conferences/past/html/index.en.html.

- Cheun, S., I. von Köppen-Mertes, and B. Weller (2009): "The Collateral Frameworks of the Eurosystem, the Federal Reserve System and the Bank of England and the Financial Market Turmoil," Occasional Paper Series No. 107, European Central Bank.
- ČIHÁK, M., T. HARJES, AND E. STAVREV (2009): "Euro Area Monetary Policy in Uncharted Waters," IMF Working Paper No. 09/185, International Monetary Fund.
- CLOUSE, J. A., AND J. P. DOW (1999): "Fixed Costs and the Behaviour of the Federal Funds Rate," *Journal of Banking and Finance*, 23(7), 1015–1029.
- ———— (2002): "A Computational Model of Banks' Optimal Reserve Management Policy," Journal of Economic Dynamics and Control, 26(11), 1787–1814.
- DISYATAT, P. (2011): "The Bank Lending Channel Revisited," Journal of Money, Credit and Banking, 43(4), 711–734.
- EISENSCHMIDT, J., A. HIRSCH, AND T. LINZERT (2009): "Bidding Behaviour in the ECB's Main Refinancing Operations During the Financial Crisis," European Central Bank Working Paper No. 1052.
- European Central Bank (2012): The Implementation of Monetary Policy in the Euro Area. General Documentation on Eurosystem Monetary Policy Instruments and Procedures. European Central Bank, Frankfurt.
- ———— (2014): "Open Market Operations: Calculation of the Benchmark Allotment Amount in the Main Refinancing Operations," 06.05.2014, https://www.ecb.europa.eu/mopo/implement/omo/html/index.en.html.
- EWERHART, C. (2002): "A Model of the Eurosystem's Operational Framework for Monetary Policy Implementation," European Central Bank Working Paper No. 197.
- FREIXAS, X., AND J. JORGE (2008): "The Role of Interbank Markets in Monetary Policy: A Model with Rationing," *Journal of Money, Credit and Banking*, 40(6), 1151–1176.
- Furfine, C. (2004): "Standing Facilities and Interbank Borrowing: Evidence from the Fed's New Discount Window," Working Paper No. 2004-01, Federal Reserve Bank of Chicago.
- FURFINE, C. H. (2000): "Interbank Payments and the Daily Federal Funds Rate," *Journal of Monetary Economics*, 46(2), 535–553.
- Gertler, M., and S. Gilchrist (1993): "The Role of Credit Market Imperfections in the Monetary Transmission Mechanism: Arguments and Evidence," *Scandinavian Journal of Economics*, 95, 43–64.
- GOODHART, C. A. E. (2013): "The Potential Instruments of Monetary Policy," *Central Bank Review*, 13, 1–15.
- Hamilton, J. D. (1996): "The Daily Market for Federal Funds," *Journal of Political Economy*, 104(1), 26–56.
- HAUCK, A., AND U. NEYER (2013): "A Model of the Eurosystem's Operational Framework and the Euro Overnight Interbank Market," forthcoming in European Journal of Political Economy.

- Heider, F., M. Hoerova, and C. Holthausen (2009): "Liquidity Hoarding and Interbank Market Spreads: The Role of Counterparty Risk," European Central Bank Working Paper No. 1126.
- Ho, T. S. Y., and A. Saunders (1985): "A Micro Model of the Federal Funds Market," *Journal of Finance*, 40, 977–986.
- JIMÉNEZ, G., S. ONGENA, J.-L. PEYDRÓ, AND J. SAURINA (2012): "Credit Supply and Monetary Policy: Identifying the Bank Balance-Sheet Channel with Loan Applications," *American Economic Review*, 102(5), 2301–26.
- Kashyap, A. K., and J. C. Stein (1995): "The Impact of Monetary Policy on Bank Balance Sheets," Carnegie-Rochester Conference Series on Public Policy, 42, 151–195.
- LENZA, M., H. PILL, AND L. REICHLIN (2010): "Monetary Policy in Exceptional Times," Economic Policy, 25(62), 295–339.
- Nautz, D. (1998): "Banks' Demand for Reserves When Future Monetary Policy is Uncertain," *Journal of Monetary Economics*, 42(1), 161–183.
- NAUTZ, D., AND J. OECHSSLER (2003): "The Repo Auctions of the European Central Bank and the Vanishing Quota Puzzle," *Scandinavian Journal of Economics*, 105(2), 207–220.
- ———— (2006): "Overbidding in Fixed Rate Tenders An Empirical Assessment of Alternative Explanations," *European Economic Review*, 50(3), 631–646.
- NEYER, U. (2009): "Interest on Reserves and the Flexibility of Monetary Policy in the Euro Area," Scandinavian Journal of Economics, 111(2), 417–438.
- NEYER, U., AND J. WIEMERS (2004): "The Influence of a Heterogeneous Banking Sector on the Interbank Market Rate in the Euro Area," Swiss Journal of Economics and Statistics, 140(3), 395–428.
- PEEK, J., AND E. S. ROSENGREEN (2012): "The Role of Banks in the Transmission of Monetary Policy," in *The Oxford Handbook of Banking*, ed. by A. N. Berger, P. Molyneux, and J. O. Wilson, pp. 257–277, Oxford. Oxford University Press.
- PÉREZ-QUIRÓS, G., AND H. RODRÍGUEZ-MENDIZÁBAL (2006): "The Daily Market for Funds in Europe: What Has Changed with the EMU?," *Journal of Money, Banking, and Credit*, 38(1), 91–110.
- VÄLIMÄKI, T. (2001): "Fixed Rate Tenders and the Overnight Money Market Equilibrium," Bank of Finland Discussion Paper No. 8/2001.
- Whitesell, W. (2006): "Interest Rate Corridors and Reserves," *Journal of Monetary Economics*, 53(6), 1177–1195.

PREVIOUS DISCUSSION PAPERS

- Hottenrott, Hanna and Lopes-Bento, Cindy, R&D Partnerships and Innovation Performance: Can There be too Much of a Good Thing?, July 2014.
- Hottenrott, Hanna and Lawson, Cornelia, Flying the Nest: How the Home Department Shapes Researchers' Career Paths, July 2014.
- Hottenrott, Hanna, Lopes-Bento, Cindy and Veugelers, Reinhilde, Direct and Cross-Scheme Effects in a Research and Development Subsidy Program, July 2014.
- Dewenter, Ralf and Heimeshoff, Ulrich, Do Expert Reviews Really Drive Demand? Evidence from a German Car Magazine, July 2014.
- Bataille, Marc, Steinmetz, Alexander and Thorwarth, Susanne, Screening Instruments for Monitoring Market Power in Wholesale Electricity Markets Lessons from Applications in Germany, July 2014.
- 149 Kholodilin, Konstantin A., Thomas, Tobias and Ulbricht, Dirk, Do Media Data Help to Predict German Industrial Production?, July 2014.
- Hogrefe, Jan and Wrona, Jens, Trade, Tasks, and Trading: The Effect of Offshoring on Individual Skill Upgrading, June 2014.
- 147 Gaudin, Germain and White, Alexander, On the Antitrust Economics of the Electronic Books Industry, May 2014.
- Alipranti, Maria, Milliou, Chrysovalantou and Petrakis, Emmanuel, Price vs. Quantity Competition in a Vertically Related Market, May 2014. Published in: Economics Letters, 124 (1) (2014), pp.122-126.
- Blanco, Mariana, Engelmann, Dirk, Koch, Alexander K., and Normann, Hans-Theo, Preferences and Beliefs in a Sequential Social Dilemma: A Within-Subjects Analysis, May 2014.
 Published in: Games and Economic Behavior, 87 (2014), pp.122-135.
- 144 Jeitschko, Thomas D., Jung, Yeonjei and Kim, Jaesoo, Bundling and Joint Marketing by Rival Firms, May 2014.
- 143 Benndorf, Volker and Normann, Hans-Theo, The Willingness to Sell Private Data, May 2014.
- Dauth, Wolfgang and Suedekum, Jens, Globalization and Local Profiles of Economic Growth and Industrial Change, April 2014.
- Nowak, Verena, Schwarz, Christian and Suedekum, Jens, Asymmetric Spiders: Supplier Heterogeneity and the Organization of Firms, April 2014.
- Hasnas, Irina, A Note on Consumer Flexibility, Data Quality and Collusion, April 2014.
- Baye, Irina and Hasnas, Irina, Consumer Flexibility, Data Quality and Location Choice, April 2014.
- 138 Aghadadashli, Hamid and Wey, Christian, Multi-Union Bargaining: Tariff Plurality and Tariff Competition, April 2014.

- Duso, Tomaso, Herr, Annika and Suppliet, Moritz, The Welfare Impact of Parallel Imports: A Structural Approach Applied to the German Market for Oral Anti-diabetics, April 2014.

 Forthcoming in: Health Economics.
- Haucap, Justus and Müller, Andrea, Why are Economists so Different? Nature, Nurture and Gender Effects in a Simple Trust Game, March 2014.
- Normann, Hans-Theo and Rau, Holger A., Simultaneous and Sequential Contributions to Step-Level Public Goods: One vs. Two Provision Levels, March 2014.

 Forthcoming in: Journal of Conflict Resolution.
- Bucher, Monika, Hauck, Achim and Neyer, Ulrike, Frictions in the Interbank Market and Uncertain Liquidity Needs: Implications for Monetary Policy Implementation, July 2014 (First Version March 2014).
- 133 Czarnitzki, Dirk, Hall, Bronwyn, H. and Hottenrott, Hanna, Patents as Quality Signals? The Implications for Financing Constraints on R&D?, February 2014.
- Dewenter, Ralf and Heimeshoff, Ulrich, Media Bias and Advertising: Evidence from a German Car Magazine, February 2014.
 Forthcoming in: Review of Economics.
- Baye, Irina and Sapi, Geza, Targeted Pricing, Consumer Myopia and Investment in Customer-Tracking Technology, February 2014.
- 130 Clemens, Georg and Rau, Holger A., Do Leniency Policies Facilitate Collusion? Experimental Evidence, January 2014.
- Hottenrott, Hanna and Lawson, Cornelia, Fishing for Complementarities: Competitive Research Funding and Research Productivity, December 2013.
- Hottenrott, Hanna and Rexhäuser, Sascha, Policy-Induced Environmental Technology and Inventive Efforts: Is There a Crowding Out?, December 2013.
- Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, The Rise of the East and the Far East: German Labor Markets and Trade Integration, December 2013. Forthcoming in: Journal of European Economic Association.
- Wenzel, Tobias, Consumer Myopia, Competition and the Incentives to Unshroud
 Add-on Information, December 2013.
 Published in: Journal of Economic Behavior and Organization, 98 (2014), pp. 89-96.
- Schwarz, Christian and Suedekum, Jens, Global Sourcing of Complex Production Processes, December 2013.
 Published in: Journal of International Economics, 93 (2014), pp. 123-139.
- Defever, Fabrice and Suedekum, Jens, Financial Liberalization and the Relationship-Specificity of Exports, December 2013. Published in: Economics Letters, 122 (2014), pp. 375-379.
- Bauernschuster, Stefan, Falck, Oliver, Heblich, Stephan and Suedekum, Jens,
 Why Are Educated and Risk-Loving Persons More Mobile Across Regions?,
 December 2013.
 Published in: Journal of Economic Behavior and Organization, 98 (2014), pp. 56-69.
- Hottenrott, Hanna and Lopes-Bento, Cindy, Quantity or Quality? Knowledge Alliances and their Effects on Patenting, December 2013.

- Hottenrott, Hanna and Lopes-Bento, Cindy, (International) R&D Collaboration and SMEs: The Effectiveness of Targeted Public R&D Support Schemes, December 2013.
 Forthcoming in: Research Policy.
- Giesen, Kristian and Suedekum, Jens, City Age and City Size, November 2013. Forthcoming in: European Economic Review.
- 119 Trax, Michaela, Brunow, Stephan and Suedekum, Jens, Cultural Diversity and Plant-Level Productivity, November 2013.
- Manasakis, Constantine and Vlassis, Minas, Downstream Mode of Competition With Upstream Market Power, November 2013.
 Published in: Research in Economics, 68 (2014), pp. 84-93.
- 117 Sapi, Geza and Suleymanova, Irina, Consumer Flexibility, Data Quality and Targeted Pricing, November 2013.
- Hinloopen, Jeroen, Müller, Wieland and Normann, Hans-Theo, Output Commitment Through Product Bundling: Experimental Evidence, November 2013. Published in: European Economic Review 65 (2014), pp. 164-180.
- 115 Baumann, Florian, Denter, Philipp and Friehe Tim, Hide or Show? Endogenous Observability of Private Precautions Against Crime When Property Value is Private Information, November 2013.
- Fan, Ying, Kühn, Kai-Uwe and Lafontaine, Francine, Financial Constraints and Moral Hazard: The Case of Franchising, November 2013.
- Aguzzoni, Luca, Argentesi, Elena, Buccirossi, Paolo, Ciari, Lorenzo, Duso, Tomaso, Tognoni, Massimo and Vitale, Cristiana, They Played the Merger Game: A Retrospective Analysis in the UK Videogames Market, October 2013. Forthcoming in: Journal of Competition and Economics under the title: "A Retrospective Merger Analysis in the UK Videogames Market".
- Myrseth, Kristian Ove R., Riener, Gerhard and Wollbrant, Conny, Tangible Temptation in the Social Dilemma: Cash, Cooperation, and Self-Control, October 2013.
- Hasnas, Irina, Lambertini, Luca and Palestini, Arsen, Open Innovation in a Dynamic Cournot Duopoly, October 2013.
 Published in: Economic Modelling, 36 (2014), pp. 79-87.
- Baumann, Florian and Friehe, Tim, Competitive Pressure and Corporate Crime, September 2013.
- Böckers, Veit, Haucap, Justus and Heimeshoff, Ulrich, Benefits of an Integrated European Electricity Market, September 2013.
- Normann, Hans-Theo and Tan, Elaine S., Effects of Different Cartel Policies: Evidence from the German Power-Cable Industry, September 2013. Forthcoming in: Industrial and Corporate Change.
- Haucap, Justus, Heimeshoff, Ulrich, Klein, Gordon J., Rickert, Dennis and Wey, Christian, Bargaining Power in Manufacturer-Retailer Relationships, September 2013.
- 106 Baumann, Florian and Friehe, Tim, Design Standards and Technology Adoption: Welfare Effects of Increasing Environmental Fines when the Number of Firms is Endogenous, September 2013.

- Jeitschko, Thomas D., NYSE Changing Hands: Antitrust and Attempted Acquisitions of an Erstwhile Monopoly, August 2013.
 Published in: Journal of Stock and Forex Trading, 2 (2) (2013), pp. 1-6.
- Böckers, Veit, Giessing, Leonie and Rösch, Jürgen, The Green Game Changer: An Empirical Assessment of the Effects of Wind and Solar Power on the Merit Order, August 2013.
- Haucap, Justus and Muck, Johannes, What Drives the Relevance and Reputation of Economics Journals? An Update from a Survey among Economists, August 2013.
- Jovanovic, Dragan and Wey, Christian, Passive Partial Ownership, Sneaky Takeovers, and Merger Control, August 2013.
 Revised version forthcoming in: Economics Letters.
- Haucap, Justus, Heimeshoff, Ulrich, Klein, Gordon J., Rickert, Dennis and Wey, Christian, Inter-Format Competition Among Retailers – The Role of Private Label Products in Market Delineation, August 2013.
- Normann, Hans-Theo, Requate, Till and Waichman, Israel, Do Short-Term Laboratory Experiments Provide Valid Descriptions of Long-Term Economic Interactions? A Study of Cournot Markets, July 2013. Forthcoming in: Experimental Economics.
- 99 Dertwinkel-Kalt, Markus, Haucap, Justus and Wey, Christian, Input Price Discrimination (Bans), Entry and Welfare, June 2013.
- Aguzzoni, Luca, Argentesi, Elena, Ciari, Lorenzo, Duso, Tomaso and Tognoni, Massimo, Ex-post Merger Evaluation in the UK Retail Market for Books, June 2013.
- Oaprice, Stéphane and von Schlippenbach, Vanessa, One-Stop Shopping as a Cause of Slotting Fees: A Rent-Shifting Mechanism, May 2012.

 Published in: Journal of Economics and Management Strategy, 22 (2013), pp. 468-487.
- Wenzel, Tobias, Independent Service Operators in ATM Markets, June 2013. Published in: Scottish Journal of Political Economy, 61 (2014), pp. 26-47.
- 95 Coublucq, Daniel, Econometric Analysis of Productivity with Measurement Error: Empirical Application to the US Railroad Industry, June 2013.
- Oublucq, Daniel, Demand Estimation with Selection Bias: A Dynamic Game Approach with an Application to the US Railroad Industry, June 2013.
- 93 Baumann, Florian and Friehe, Tim, Status Concerns as a Motive for Crime?, April 2013.
- Jeitschko, Thomas D. and Zhang, Nanyun, Adverse Effects of Patent Pooling on Product Development and Commercialization, April 2013.
 Published in: The B. E. Journal of Theoretical Economics, 14 (1) (2014), Art. No. 2013-0038.
- 91 Baumann, Florian and Friehe, Tim, Private Protection Against Crime when Property Value is Private Information, April 2013.
 Published in: International Review of Law and Economics, 35 (2013), pp. 73-79.
- Baumann, Florian and Friehe, Tim, Cheap Talk About the Detection Probability,
 April 2013.
 Published in: International Game Theory Review, 15 (2013), Art. No. 1350003.
- Pagel, Beatrice and Wey, Christian, How to Counter Union Power? Equilibrium Mergers in International Oligopoly, April 2013.

- Jovanovic, Dragan, Mergers, Managerial Incentives, and Efficiencies, April 2014 (First Version April 2013).
- Heimeshoff, Ulrich and Klein Gordon J., Bargaining Power and Local Heroes, March 2013.
- Bertschek, Irene, Cerquera, Daniel and Klein, Gordon J., More Bits More Bucks? Measuring the Impact of Broadband Internet on Firm Performance, February 2013. Published in: Information Economics and Policy, 25 (2013), pp. 190-203.
- Rasch, Alexander and Wenzel, Tobias, Piracy in a Two-Sided Software Market, February 2013.
 Published in: Journal of Economic Behavior & Organization, 88 (2013), pp. 78-89.
- Bataille, Marc and Steinmetz, Alexander, Intermodal Competition on Some Routes in Transportation Networks: The Case of Inter Urban Buses and Railways, January 2013.
- Haucap, Justus and Heimeshoff, Ulrich, Google, Facebook, Amazon, eBay: Is the Internet Driving Competition or Market Monopolization?, January 2013. Published in: International Economics and Economic Policy, 11 (2014), pp. 49-61.
- Regner, Tobias and Riener, Gerhard, Voluntary Payments, Privacy and Social Pressure on the Internet: A Natural Field Experiment, December 2012.
- Dertwinkel-Kalt, Markus and Wey, Christian, The Effects of Remedies on Merger Activity in Oligopoly, December 2012.
- Baumann, Florian and Friehe, Tim, Optimal Damages Multipliers in Oligopolistic Markets, December 2012.
- Duso, Tomaso, Röller, Lars-Hendrik and Seldeslachts, Jo, Collusion through Joint R&D: An Empirical Assessment, December 2012.
 Forthcoming in: The Review of Economics and Statistics.
- Baumann, Florian and Heine, Klaus, Innovation, Tort Law, and Competition,
 December 2012.
 Published in: Journal of Institutional and Theoretical Economics, 169 (2013), pp. 703-719.
- Coenen, Michael and Jovanovic, Dragan, Investment Behavior in a Constrained Dictator Game, November 2012.
- Gu, Yiquan and Wenzel, Tobias, Strategic Obfuscation and Consumer Protection Policy in Financial Markets: Theory and Experimental Evidence, November 2012. Forthcoming in: Journal of Industrial Economics under the title "Strategic Obfuscation and Consumer Protection Policy".
- Haucap, Justus, Heimeshoff, Ulrich and Jovanovic, Dragan, Competition in Germany's Minute Reserve Power Market: An Econometric Analysis, November 2012.
 Published in: The Energy Journal, 35 (2014), pp. 139-158.
- Normann, Hans-Theo, Rösch, Jürgen and Schultz, Luis Manuel, Do Buyer Groups Facilitate Collusion?, November 2012.
- Riener, Gerhard and Wiederhold, Simon, Heterogeneous Treatment Effects in Groups, November 2012.
 Published in: Economics Letters, 120 (2013), pp 408-412.

- Berlemann, Michael and Haucap, Justus, Which Factors Drive the Decision to Boycott and Opt Out of Research Rankings? A Note, November 2012.
- Muck, Johannes and Heimeshoff, Ulrich, First Mover Advantages in Mobile Telecommunications: Evidence from OECD Countries, October 2012.
- Karaçuka, Mehmet, Çatik, A. Nazif and Haucap, Justus, Consumer Choice and Local Network Effects in Mobile Telecommunications in Turkey, October 2012. Published in: Telecommunications Policy, 37 (2013), pp. 334-344.
- 69 Clemens, Georg and Rau, Holger A., Rebels without a Clue? Experimental Evidence on Partial Cartels, April 2013 (First Version October 2012).
- Regner, Tobias and Riener, Gerhard, Motivational Cherry Picking, September 2012.
- Fonseca, Miguel A. and Normann, Hans-Theo, Excess Capacity and Pricing in Bertrand-Edgeworth Markets: Experimental Evidence, September 2012. Published in: Journal of Institutional and Theoretical Economics, 169 (2013), pp. 199-228.
- Riener, Gerhard and Wiederhold, Simon, Team Building and Hidden Costs of Control, September 2012.
- Fonseca, Miguel A. and Normann, Hans-Theo, Explicit vs. Tacit Collusion The Impact of Communication in Oligopoly Experiments, August 2012. Published in: European Economic Review, 56 (2012), pp. 1759-1772.
- Jovanovic, Dragan and Wey, Christian, An Equilibrium Analysis of Efficiency Gains from Mergers, July 2012.
- Dewenter, Ralf, Jaschinski, Thomas and Kuchinke, Björn A., Hospital Market Concentration and Discrimination of Patients, July 2012. Published in: Schmollers Jahrbuch, 133 (2013), pp. 345-374.
- Von Schlippenbach, Vanessa and Teichmann, Isabel, The Strategic Use of Private Quality Standards in Food Supply Chains, May 2012.
 Published in: American Journal of Agricultural Economics, 94 (2012), pp. 1189-1201.
- 61 Sapi, Geza, Bargaining, Vertical Mergers and Entry, July 2012.
- Jentzsch, Nicola, Sapi, Geza and Suleymanova, Irina, Targeted Pricing and Customer Data Sharing Among Rivals, July 2012.
 Published in: International Journal of Industrial Organization, 31 (2013), pp. 131-144.
- Lambarraa, Fatima and Riener, Gerhard, On the Norms of Charitable Giving in Islam: A Field Experiment, June 2012.
- Duso, Tomaso, Gugler, Klaus and Szücs, Florian, An Empirical Assessment of the 2004 EU Merger Policy Reform, June 2012. Published in: Economic Journal, 123 (2013), F596-F619.
- Dewenter, Ralf and Heimeshoff, Ulrich, More Ads, More Revs? Is there a Media Bias in the Likelihood to be Reviewed?, June 2012.
- Böckers, Veit, Heimeshoff, Ulrich and Müller Andrea, Pull-Forward Effects in the German Car Scrappage Scheme: A Time Series Approach, June 2012.
- Kellner, Christian and Riener, Gerhard, The Effect of Ambiguity Aversion on Reward Scheme Choice, June 2012.

- De Silva, Dakshina G., Kosmopoulou, Georgia, Pagel, Beatrice and Peeters, Ronald, The Impact of Timing on Bidding Behavior in Procurement Auctions of Contracts with Private Costs, June 2012.

 Published in: Review of Industrial Organization, 41 (2013), pp.321-343.
- Benndorf, Volker and Rau, Holger A., Competition in the Workplace: An Experimental Investigation, May 2012.
- Haucap, Justus and Klein, Gordon J., How Regulation Affects Network and Service Quality in Related Markets, May 2012.
 Published in: Economics Letters, 117 (2012), pp. 521-524.
- Dewenter, Ralf and Heimeshoff, Ulrich, Less Pain at the Pump? The Effects of Regulatory Interventions in Retail Gasoline Markets, May 2012.
- Böckers, Veit and Heimeshoff, Ulrich, The Extent of European Power Markets, April 2012.
- 49 Barth, Anne-Kathrin and Heimeshoff, Ulrich, How Large is the Magnitude of Fixed-Mobile Call Substitution? Empirical Evidence from 16 European Countries, April 2012.

 Forthcoming in: Telecommunications Policy.
- Herr, Annika and Suppliet, Moritz, Pharmaceutical Prices under Regulation: Tiered Co-payments and Reference Pricing in Germany, April 2012.
- Haucap, Justus and Müller, Hans Christian, The Effects of Gasoline Price Regulations: Experimental Evidence, April 2012.
- Stühmeier, Torben, Roaming and Investments in the Mobile Internet Market, March 2012.
 Published in: Telecommunications Policy, 36 (2012), pp. 595-607.
- Graf, Julia, The Effects of Rebate Contracts on the Health Care System, March 2012, Published in: The European Journal of Health Economics, 15 (2014), pp.477-487.
- Pagel, Beatrice and Wey, Christian, Unionization Structures in International Oligopoly,
 February 2012.
 Published in: Labour: Review of Labour Economics and Industrial Relations, 27 (2013),
 pp. 1-17.
- Gu, Yiquan and Wenzel, Tobias, Price-Dependent Demand in Spatial Models, January 2012.
 Published in: B. E. Journal of Economic Analysis & Policy, 12 (2012), Article 6.
- Barth, Anne-Kathrin and Heimeshoff, Ulrich, Does the Growth of Mobile Markets Cause the Demise of Fixed Networks? Evidence from the European Union, January 2012.

 Forthcoming in: Telecommunications Policy.
- 41 Stühmeier, Torben and Wenzel, Tobias, Regulating Advertising in the Presence of Public Service Broadcasting, January 2012.
 Published in: Review of Network Economics, 11/2 (2012), Article 1.

Older discussion papers can be found online at: http://ideas.repec.org/s/zbw/dicedp.html

Heinrich-Heine-University of Düsseldorf Department of Economics

Universitätsstraße 1 _ 40225 Düsseldorf www.dice.hhu.de